

Temi di Discussione

(Working Papers)

Pairwise trading in the money market during the European sovereign debt crisis

by Edoardo Rainone

I1160



Temi di discussione

(Working papers)

Pairwise trading in the money market during the European sovereign debt crisis

by Edoardo Rainone

Number 1160 - December 2017

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Ines Buono, Marco Casiraghi, Valentina Aprigliano, Nicola Branzoli, Francesco Caprioli, Emanuele Ciani, Vincenzo Cuciniello, Davide Delle Monache, Giuseppe Ilardi, Andrea Linarello, Juho Taneli Makinen, Valerio Nispi Landi, Lucia Paola Maria Rizzica, Massimiliano Stacchini. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

PAIRWISE TRADING IN THE MONEY MARKET DURING THE EUROPEAN SOVEREIGN DEBT CRISIS

by Edoardo Rainone*

Abstract

This paper studies over-the-counter (OTC) trading in the unsecured interbank market for euro funds. The goal of our analysis is to identify the determinants of the probability of trading, the bilateral rate, and the quantity exchanged during the European sovereign debt crisis. We show how the specific features of this market bring to a non-standard estimation framework. We propose a dyadic econometric model with shadow rates to control for potentially endogenous matching with the counterparty, and construct a unique dataset containing banks' characteristics and bilateral trades to study trading patterns. The estimates provide mild evidence towards the existence of shadow rates. Active monitoring decreased market access to low equity and illiquid borrowers, while dispersion in rates and quantities was mainly driven by banks' nationality, especially at the peak of the crisis.

JEL Classification: E50, E40, C30, G01, G10, D40.

Keywords: interbank networks, payment systems, sample selection models, two-step estimation, over-the-counter market, money, dyadic model, financial crisis.

Contents

Introduction	5
Related Literature	7
Aggregate Evidence	7
A Decentralized Market with Counterparty-risk Uncertainty and Risk-free	
Counterparty of Last Resort	
A Dyadic Econometric Model with Shadow Rates	9
Empirical Analysis	
Concluding Remarks	
erences	
les and figures	
pendix	
	Introduction

^{*}Bank of Italy, DG for Markets and Payment Systems - Payment System Directorate. E-mail: edoardo.rainone@bancaditalia.it.

1 Introduction¹

Before the recent financial crises the unsecured money market was the most important channel to reallocate liquidity among the banks. During the crises the interbank markets were remarkably stressed. Given the importance of these OTC markets, the fact drew the attention of many policy makers and researchers.

A large number of theories have been proposed to explain the features of bilateral trades in OTC markets (see Afonso and Lagos, 2015; Bech and Monnet, 2016; Blasques et al., 2016; Duffie et al., 2005, among the others), but the empirical literature still lacks in providing econometric models and evidences to better understand these pairwise outcomes. While there are empirical studies investigating interbank markets after the 2008 crisis (Afonso et al., 2011; Angelini et al., 2011), few evidences about the European sovereign crisis are available in the literature.

In this paper we contribute in both these directions. Our main goal is to empirically study the evolution of bilateral trading outcomes in the unsecured interbank market for euro funds during the European sovereign debt crisis, a task never explored in the literature. More specifically, we want to understand how banks characteristics affect the probability to trade, bilateral rates and quantities. We are not aware of any study that analyzes formally and empirically this topic.² Evidences from our analysis can be used to assess European market fragmentation (de Andoain et al., 2014; Mayordomo et al., 2015), to explain rate dispersion (Gaspar et al., 2008) or to study supply concentration, preventing smooth and homogeneous pass-through of monetary policy.

To get to this point, the preliminary questions we have to answer are: how can we consistently estimate the effects of banks characteristics on such outcomes, namely the bilateral rate, the quantity and the probability of trading? Can we use a standard econometric model? Loan's rate and quantity are only observed when that specific pair of borrower and lender agree on a bilateral negotiation. Given the decentralized nature of the market, participants really pick up the phone and call each other to set up the loan and bargain prices (Afonso and Lagos, 2015).³ In this sense, a matching model is a possible framework for studying the equilib-

²Angelini et al. (2011) is the only study using pairwise data we are aware of. They analyzed the impact of the subprime crisis on the trades of the Italian platform e-MID. Frutos et al. (2016) describe the interbank market during the European sovereign crisis, not modeling bilateral outcomes nor studying the effect of bank characteristics and nationality.

³This feature characterizes also other type of decentralized markets.

¹Bank of Italy, DG for Markets and Payment Systems - Payment System Directorate. I thank Massimiliano Affinito, Tiziano Arduini, Olivier Armantier, Carlo Del Bello, Silvia Gabrieli, Co-pierre Georg, Helina Laakkonen, Fabrizio Mattesini, James McAndrews, Hector Perez Saiz, Francisco Rivadeneyra, Paolo Vitale, Yu Zhu, and two anonymous referees of the Bank of Italy *Temi di discussione* for their comments as well as participants in the 14th BoF-PSS Simulator seminar, the 4th Workshop in Macro Banking and Finance, the Bank of Canada and Payments Canada Workshop on the Modeling and Simulation of Payments and Other Financial System Infrastructures, The 2017 RCEA Macro-Money-Finance Workshop, The Banque de France lunch seminar and the Economics of Payments VIII. I wish to thank Giovanni di Iasio, Marco Rocco and Francesco Vacirca for sharing data and thoughts with me, Salvatore Alonzo and Fabrizio Palmisani for giving me the opportunity to investigate this topic. All the errors are my own. The author of this paper is member of one of the user groups with access to TARGET2 data in accordance with Article 1(2) of Decision ECB/2010/9 of 29 July 2010 on access to and use of certain TARGET2 data. The Bank of Italy and the PSSC have checked the paper against the rules for guaranteeing the confidentiality of transaction-level data imposed by the PSSC pursuant to Article 1(4) of the above mentioned issue. The views expressed in the paper are solely those of the author and do not necessarily represent the views of the Eurosystem or of the Bank of Italy.

rium formation of these relationships (Chiappori and Salanié, 2016; Fox, 2009; Graham, 2011; Sørensen, 2007). It implies that the trading patterns can follow rules that are difficult to be observed by the econometrician.⁴ The endogenous matching process, generates a counterparty selection bias, and can be seen as a specification error in the spirit of Heckman (1979).⁵ We show that the role played by money market-specific unobservable factors (such as monitoring and searching costs, see Afonso and Lagos, 2015; Blasques et al., 2016) and the presence of the central bank as a lender of last resort lead to a non-standard estimation framework that departs from a classic dyadic econometric model (Cameron and Miller, 2014; Kenny et al., 2006). To solve this issue we apply a control function approach to account for the selection bias. More precisely, the solution proposed in this paper is a new dyadic econometric model with shadow rates. The concept of shadow rates is used to model such selectivity issues and to capture the unique features of this market, as unobservable searching and monitoring costs (Blasques et al., 2016) or endogenous intermediation (Babus and Hu, 2017). In developing our econometric model, we discuss the potential bias resulting from not simultaneously modeling the matching process when bilateral rates and quantities are studied.

With the proposed econometric model at hand, we study the unsecured money market for euro funds during the European sovereign crisis, using a unique dataset containing the characteristics of banks operating worldwide and bilateral trades. To te best of our knowledge, this is the first attempt to jointly analyze the information from transaction-level data and characteristics of global banks operating in euro. We find a remarkable dispersion in rates and quantities driven by banks nationality and balance sheet composition, especially during the peaks of the crisis. More specifically, we witness an important role played by borrower characteristics. Balance sheet composition and nationality impact dramatically on the probability of borrowing money in general and especially at low rates. Most notably, bank's nationality, equity and size played a dominant role in determining access to the market and lower rates, which is coherent with a credit-risk story and an active monitoring by the lenders. Lender characteristics matter as well, especially in explaining the quantity of liquidity supplied in the market -which can be seen as liquidity hoarding- Heider et al. (2015). Interestingly, we find substantial time variation of these effects and differential magnitudes across countries during the crisis. Among the many new evidences collected, we found that Italian and Spanish borrowers paid increasingly higher spreads from the first phase of the sovereign crisis (April 2010) towards the second (August 2011). On the other side of the market, after the second phase of the crisis, banks from some of the most stressed countries (namely Italy, Spain and Greece) lent at extremely higher rates, because of the sudden scarcity of liquidity. After the first LTRO such spreads were cleared from the market by the huge amount of liquidity provided by the Eurosystem. A detailed description of the main findings is provided in Section 6.

The rest of the paper is organized as follows. Section 2 briefly connects this research with the related literature. Section 3 describes some aggregate evidence that motivates a pairwise analysis. Section 4 presents a conceptual economic framework for a decentralized unsecured money market. Section 5 outlines the proposed dyadic econometric model and the concept of

⁴Some of those patterns have been recently studied. Among the others, Affinito (2012) and Cocco et al. (2009) investigate the role of relationship lending, Rainone (2015) and Gabrieli and Georg (2014) study the role of the network structures.

⁵For example, if unobservable variables determine both the probability that two bank get in touch and the rate (quantity) they agree (exchange), then the estimated parameters of loans' outcomes -i.e. rate and quantity- can be biased as well. This generates a sample selection bias with the implication that simple OLS estimation of the loan rate and quantity functions is not consistent.

shadow rates. Section 6 describe the data, the specification and the results of the empirical analysis, Section 7 concludes.

2 Related Literature

In doing this exercise, we are bridging the literature on the role of liquidity hoarding and counterparty risk in interbank markets (as Afonso et al., 2011; Angelini et al., 2011; Heider et al., 2015, among the others) with the theoretical literature focused on explaining the features of OTC markets (like Afonso and Lagos, 2015; Bech and Monnet, 2016; Blasques et al., 2016; Duffie et al., 2005, among the others). Liquidity hoarding and counterparty credit risk have been identified as the main channels for idiosyncratic shocks propagation and systemic reduction of liquidity (Afonso et al., 2011; Angelini et al., 2011; Heider et al., 2015). When strong uncertainty on future own and others' liquidity condition occurs, banks can decide to hoard liquidity to prevent future shocks and may perceive some counterparties as excessively risky Heider et al. (2015). Caballero and Krishnamurthy (2008) used Knightian uncertainty (Knight, 2012) to explain market-wide capital immobility and liquidity hoarding. In their model agents focus on the worst case scenario and become self protective.⁶ Among the others, Acharya and Skeie (2011) proposed a model for liquidity hoarding in which a reduction in quantities and an increase in prices is also driven by lenders' characteristics and not only by borrowers' ones.⁷ They also highlight the lack in empirical works that jointly look at prices and quantities in the interbank market. Regarding the counterparty risk, as Afonso et al. (2011) pointed out, many theoretical models focused on adverse selection and inability of lenders to discern good from bad banks, Flannery (1996) is an example. On the other hand, some banks can be excluded from the market because they are seen as too risky from the others, see Furfine (2001) among the others.

3 Aggregate Evidence

As discussed above, great attention was paid to the variation of money market aggregate outcomes during the recent financial crises. Figure 1 reports the total number of bilateral trades, the total value of loans and the average rate in the unsecured money market for euro funds from may 2008 to the end of 2012.⁸ The decrease of interbank trades and quantity exchanged from the subprime crisis is reported respectively in panel (a) and (b). The evolution of the market rate is depicted in panel (c). During the time span considered, large variations are observed in these plots, reflecting many episodes and events. Such macro picture can tell

⁶In Caballero and Krishnamurthy a lender of last resort can be beneficial to let the agents to free capital and waste less private liquidity. At the same time interventions must not be too frequent because of a moral hazard problem. They highlight that uncertainty is particularly strong when "new" shocks occur, thus no historical information is available to agents. The subprime crisis and the European sovereign crisis were new in this sense. Regarding the latter, country specific crises were observed in the past, but it was the first time in a context of a single currency union where the break-up scenario might have occurred.

⁷The rollover risk is the key component in their model and generates a lending banks' precautionary demand for liquidity. It theoretically turns out that lenders might be incentivized to rise rates even for relatively safe borrowers. This dynamic is particularly relevant for longer term maturities, it reverts the usual concept that rates are only driven by borrower's characteristics (risk).

⁸These statistics are computed on our sample, that is described in detail in Section 6.1.

us something about what happened and give us some interpretation key if matched with news and events timeline, but may still hide some underlying information at a more disaggregated level.

To get more insights, we can drill down to the market side and individual bank level. Figure 2 reports the quantiles of the same variables computed at the bank-level separately for borrowers and lenders. Light shades track the interdecile range, dark shades depict the interquartile interval and the median is traced with a bold line. From these figures we can learn more, and see for example that in addition to an aggregate shrinking number of trades after the second phase of the sovereign crisis, there was also a sharp decrease in the concentration of lenders (panel (a)) and borrowers (panel (b)), measured by the interdecile range. From panel (c) we can see an opposite evolution for the exchanged quantity. After the long term refinancing operations (LTROs) conducted by the Eurosystem, most of the liquidity was exchanged by few lenders, probably acting as disseminators. Moving to rates (panels (e) and (f)), we can notice a remarkable increase of dispersion and skeweness over time and especially after the two peaks of the European sovereign crisis (in April 2010 and August 2011). This evidence implies that some banks paid significantly higher rates than others during the crisis. Which banks paid more? What are the determinants driving such remarkable dispersion?

In a decentralized market, the mandatory step forward to answer theses questions and learn more is drilling further down to the pair-level in order to understand the most granular market dynamics. In Section 6 we provide such answers and show all the knowledge gained exploiting the bilateral nature of these trades. To do that consistently, we first introduce a conceptual framework that gathers together all the bilateral outcomes we that want to study, and then we construct a proper econometric model tailored for the money market peculiarities.

4 A Decentralized Market with Counterparty-risk Uncertainty and Risk-free Counterparty of Last Resort

4.1 Monitoring and Searching

Let us introduce a naive model of bilateral trading in a decentralized unsecured money market with counterparty credit risk, searching and monitoring. The aim of this section is just to give an heuristic view of the drivers that generate observables and unobservables variables in an empirical model of bilateral trade outcomes. See Duffie et al. (2005), Afonso and Lagos (2015), Bech and Monnet (2016) and Blasques et al. (2016) among the others for detailed and structured description of such models. In this environment banks lend money to each other depending on their liquidity needs. Pairs of banks match bilaterally in this decentralized market and searching for a counterparty is costly. Given that banks may default, they are incentivize to monitor others' solvency status.

Suppose that the central bank sets a interest rate corridor with p_{OD} and p_{ML} be respectively the overnight deposit and marginal lending rates. If we allow both the lender and the borrower to exert efforts to find counterparties and the lender to monitor the solvency status of the borrower, we have the following payoffs:

Borrower payoff

$$\pi_b = p_{ML} - (p_{lb} + s_{b,l}) \tag{1}$$

Lender payoff

$$\pi_l = i_{lb}(PD_l(b)) - m_{l,b} - s_{l,b} - p_{OD}$$
⁽²⁾

where in equation (1) $s_{b,l}$ is the search cost paid by b to find l and p_{lb} is the rate paid by the borrower b to the lender l. In equation (2) i_{lb} is the expected profit for l on a loan to b, which differs from p_{lb} because b can default with probability PD(b) and depends on the lenderspecific estimate of such probability $\hat{PD}_l(b) = PD(b) + j(\sigma_{\nu}^l)$, where $\sigma_{\nu_b^l}$ is the variance of a lender-specific perception error ν_b^l about b solvency status and $j(\cdot)$ is a differentiable function. $m_{l,b}$ is the cost paid by l to monitor b. As in Blasques et al. (2016), let $\frac{\delta i_{lb}}{\delta \sigma_{\nu_b^l}} < 0$ and allow the

lender to invest an amount $m_{l,b}$ in monitoring b's status with $\frac{\delta \sigma_{\nu_b}^l}{\delta m_{l,b}} < 0$. $s_{l,b}$ is the search cost paid by l to find b.

4.2 Bilateral Rate and Volume

Suppose that each bank *i* receives an exogenous liquidity shock ξ_i that may represent client's payments or cash withdrawals. Observe that both the monitoring cost $(m_{i,k})$ and the searching cost $(s_{i,k})$ can be allowed to depend on ξ_i . These initial liquidity conditions determine the demand and the supply of liquidity in the market. Let

$$\tilde{p}_{lb} = argmax \ f(\pi_l, \pi_b, \mu_l, \mu_b, w_{lb}) \tag{3}$$

$$\tilde{q}_{lb} = \operatorname{argmax} h(\xi_l, \xi_b, y_{lb}) \tag{4}$$

be the Nash equilibrium interest rate and the liquidity exchanged in the bilateral trade between l and b, with the rate as a function of borrower and lender payoffs, their bargaining powers, μ_l and μ_b , and a set of observable and unobservable pair-specific characteristics, w_{lb} . The quantity exchanged is given by bilateral liquidity shocks and a set of observable and unobservable pair-specific characteristics, w_{lb} . The quantity specific characteristics, y_{lb} . $f(\cdot)$ and $h(\cdot)$ are differentiable functions, see Afonso and Lagos (2015) and Blasques et al. (2016) among the others for possible specifications of such functions.

In this paper we are interested in estimating the effect of observable characteristics, such as nationality and balance sheet composition, on these pairwise outcomes.

5 A Dyadic Econometric Model with Shadow Rates

Given that in such a market the price is not given, and it is formed at the pair-level, it can depend on counterparties characteristics, for example $\hat{PD}_l(b)$, $m_{l,b}$ or $s_{b,l}$. Suppose that the econometrician observes a set of realized loans in the market and she is interested in estimating how lender and borrower characteristics affect the observed bilateral rate

$$p_{lb} = l(x_l, x_b, q_{lb}, \beta, \alpha, \epsilon_{lb}), \tag{5}$$

where l(.) is a differentiable function, β contains the unknown parameters of the exogenous variables, α captures systematic and macroeconomic risk, q_{lb} is the quantity exchanged,⁹ ϵ_{lb} is the unobservable random component, x_b contains observable characteristics of the borrower that captures counterparty risk, while x_l includes observables characteristics of the lender that

⁹Observe that loan quantity is not meant to proxy counterparty risk.

represents her propensity to lend. Such empirical models could be used if we are interested in assessing market fragmentation, segregation or integration for instance.¹⁰ For simplicity, suppose the rate is a linear function of its arguments

$$p_{lb} = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + \epsilon_{lb}, \tag{6}$$

where $x_{lb} = h(x_l, x_b)$ is a pair-specific function of the relevant borrower and lender observable characteristics.

Without any prior knowledge of the data generation process induced by this decentralized market, equation (6) may look a standard dyadic model (Cameron and Miller, 2014; Kenny et al., 2006). Nevertheless, as described in Section 4, this rate is observed if both the lender and the borrower agree on the conditions of the loan conditional on the relevant rate bounds set by the central bank -i.e. if $\pi_l \geq 0 \cap \pi_b \geq 0$. As we show below, the mixture of all these ingredients brings to a non-standard estimation framework that departs from classic dyadic models. To embed the specific features of pairwise trading in the unsecured money market into our econometric model, we use the concept of shadow rates. Before getting through the detail of the proposed method, let us give some economic intuition. Suppose a lender views two potential borrowers as having different counterparty risk (or monitoring costs). Then the lender could have different rates at which is willing to lend. Similarly, a borrower may view two lenders as more or less relationship lenders, willing to stick to the borrower through thick and thin. It may be willing to pay more to a more faithful lender.

Let bank j have two shadow rates one as lender and one as borrower, let us call them $p_{L,jk}^*$ and $p_{B,jk}^*$ respectively, they both depend on the counterpart k through its counterparty risk, searching and monitoring costs and a set of observable and unobservable variables. To ease the notation let us omit the index k. If the bank is engaging the contract as lender, it will agree on setting up the loan only if the rate is higher or equal to its lender shadow rate, -i.e. $p_{lb} \ge p_{L,j}^*$, while, if the bank is acting as the borrower of the loan, it will agree only if the rate is lower or equal to its borrower shadow rate, -i.e. $p_{lb} \le p_{B,j}^*$. In this way, a loan between a lender l and a borrower b is observed if and only if $p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*$, so that a loan and its rate are observed if $I(p_{lb} \ge p_{L,l}^*)I(p_{B,b}^* \ge p_{lb}) = 1$. We assume that these shadow rates are functions of bank-specific and pair-specific characteristics:

$$p_{B,b}^* = l(k_b, z_{lb}, q_{lb}, \theta, u_{B,b}), \tag{7}$$

$$p_{L,l}^* = m(k_l, z_{lb}, q_{lb}, \gamma, u_{L,l}), \tag{8}$$

where $z_{lb} = g(z_l, z_b)$ is a pair-specific function of relevant borrower and lender characteristics, k_b and k_l are bank-specific characteristics, θ and γ are the parameters of those characteristics respectively in $l(\cdot)$ and $m(\cdot)$, $u_{B,b}$ and $u_{L,l}$ are bank specific unobservables.¹¹ Again for simplicity, suppose that those two functions are linear, so that

$$p_{B,b}^* = \theta_{0b} + \theta_1 z_{lb} + \theta_{2b} q_{lb} + \theta_3 k_b + u_{B,b}, \tag{9}$$

$$p_{L,l}^* = \gamma_{0l} + \gamma_1 z_{lb} + \gamma_{2l} q_{lb} + \gamma_3 k_l + u_{L,l}.$$
(10)

¹⁰This type of analysis is particularly relevant when the market includes participants from different countries, like the European money market.

¹¹These unobservables can also vary with the counterpart, thus being pair-specific. Here we assume they do not in order to keep the notation simple.

The intercept and the quantity slope are allowed to be lender (borrower) specific. Note that the loan rate and both the shadow rates are pair specific, it means that a bank is allowed to vary its shadow rates depending on the counterpart's characteristics. This also allows us to capture persistence in banking relationships (see Affinito, 2012; Cocco et al., 2009). Observe that θ_{0b} can also capture *b*-specific unobservable variables such as reserves, payments volatility and market access (or absence).

To get an additional connection to the stylized model presented in Section 4, observe that $u_{B,b}$ contains searching costs $(s_{b,l})$ if they are not observable to the econometrician. On the other side, $u_{L,l}$ can include unobservable monitoring and searching costs $(m_{l,b} \text{ and } s_{l,b})$.

Each pair of banks is thus characterized by a plausible rate-quantity region, that is the intersection between the two areas respectively upper and lower-countered by (9) and (10), see Figure 3. For example, the lender L1 in panel (a) has a tighter acceptable area (the dark blue one) w.r.t. lender L2, when the borrower is B1. According to Section 4, this can be generated by higher monitoring costs for L1. Let us call $s_b^* = p_{B,b}^* - p_{lb}$, $s_l^* = p_{lb} - p_{L,l}^*$ and $s_l = I(s_l^* \ge 0)$, $s_b = I(s_b^* \ge 0)$, the loan is agreed if and only if $s_l s_b = I(s_l^* \ge 0)I(s_b^* \ge 0) = 1$. From equations (6), (9) and (10) the loan is observed at zero quantity if

$$\begin{cases} \theta_{0b} - \beta_0 + \theta_1 z_{lb} - \beta_1 x_{lb} + \theta_3 k_b \ge v_{B,b}, \\ \beta_0 - \gamma_{0l} - \gamma_1 z_{lb} + \beta_1 x_{lb} - \gamma_3 k_l \ge v_{L,l}, \end{cases} (11)$$

where $v_{B,b} = \epsilon_{lb} - u_{B,b}$ and $v_{L,l} = u_{L,l} - \epsilon_{lb}$. Given that both are functions of p_{lb} but s_b is a function of the borrower shadow rate $(p_{B,b}^*)$ while s_l is a function of the lender shadow rate $(p_{L,l}^*)$, we can see them as two separate selection equations. Given their rate constraints, banks want to maximize the exchanged liquidity because searching for an additional counterpart is costly. The quantity of liquidity adjusts so that $p_{B,b}^* = p_{lb} = p_{L,l}^*$, then conditions (11) hold, the loan is observed and equations (6), (9) and (10) become a recursive system determining the quantity of money exchanged and the relative rate:

$$\begin{cases} q_{lb} = \zeta(\gamma_{0l} - \beta_0 + \gamma_1 z_{lb} - \beta_1 x_{lb} + \gamma_3 k_l) + \frac{u_{L,l} - \epsilon_{lb}}{(\alpha - \gamma_{2l})} \\ = \mu(-\theta_{0b} + \beta_0 - \theta_1 z_{lb} + \beta_1 x_{lb} - \theta_3 k_b) + \frac{\epsilon_{lb} - u_{B,b}}{(\theta_{2b} - \alpha)}, \\ p_{lb} = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + \epsilon_{lb}, \end{cases}$$
(12)

where $\mu = \frac{1}{(\theta_{2b}-\alpha)}$ and $\zeta = \frac{1}{(\alpha-\gamma_{2l})}$. Thus, the distributions of the disturbances of the system of equations (12) are conditional on inequalities (11). Since the same exogenous variables appear in conditions (11) and equations (12), the moments of these conditional distributions depend on the values of the exogenous variables. It turns out that the regressors in the system of equations (12) can be correlated with the disturbances and using ordinary least squares doesn't guarantee unbiased and consistent estimates of the parameters in the system of equations (12). However, parameters in (12) can be estimated controlling for the dependence between the disturbances, using the relationship between conditional and unconditional distributions.¹² The joint distribution of observed rates and quantities is

$$f(p_{lb}, q_{lb} | p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*) = \frac{g(p_{lb}, q_{lb})}{P([p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*])}.$$
(13)

 $^{^{12}}$ See Heckman (1974) for a detailed discussion.

where $g(p_{lb}, q_{lb})$ is the unconditional joint distribution of rates and quantities, $P([p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*])$ is the probability to observe the loan -i.e. that loan rate is included in the shadow rates interval- and $f(\cdot)$ is the conditional distribution. It implies that the likelihood function of the entire sample (including observed and unobserved loans) is

$$L = \prod_{lb \in O} f(p_{lb}, q_{lb} | p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*) P([p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*])$$
(14)

$$\times \prod_{lb \in U} P([p_{lb} \ge p_{B,b}^*, p_{lb} \le p_{L,l}^*])$$

$$= \prod_{lb \in O} g(p_{lb}, q_{lb}) \times \prod_{lb \in U} P([p_{lb} \ge p_{B,b}^*, p_{lb} \le p_{L,l}^*]).$$

Where O and U indicate the observed and unobserved partitions respectively. Observe that here the likelihood is the product of a sequence of bilateral outcomes. Modeling all jointly would hamper the treatability of our framework and the computational feasibility of the likelihood. Nevertheless, if z_b , x_b and k_b do a good job in approximate counterparty risk and z_l , x_l and k_l capture the risk of lender's portfolio correctly, our bilateral shadow rates framework can control for integrated portfolio decisions. We are pretty confident that the wide set of controls described in Section 6.2, which includes balance sheet composition, nationality and banks' activity in the market, can treat this issue effectively. From the previous derivations we can summarize our empirical model with the following system

$$p_{lb} = p_{lb}^* s_l s_b,$$

$$p_{lb}^* = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + \epsilon_{lb},$$

$$s_l = I(s_l^* \ge 0),$$

$$s_b = I(s_b^* \ge 0),$$

$$s_l^* = \omega r_l + v_{L,l},$$

$$s_b^* = \lambda r_b + v_{B,b},$$

$$(\epsilon_{lb}, v_B, v_L) \sim f\left(\begin{bmatrix} 0\\0\\0\\0\end{bmatrix}, \begin{bmatrix} \sigma_{\epsilon} & \sigma_{\epsilon v_B} & \sigma_{\epsilon v_L}\\\sigma_{\epsilon v_B} & \sigma_{v_B} & \sigma_{v_B v_L}\\\sigma_{\epsilon v_L} & \sigma_{v_B v_L} & \sigma_{v_L} \end{bmatrix}\right),$$

$$(15)$$

where f(m, V) is a trivariate density function with mean m and variance-covariance matrix V, $\omega = [\beta_0, -\gamma_{0l}, -\gamma_1, \beta_1, \alpha, -\gamma_{2l}, -\gamma_3], \kappa = [\theta_{0b}, -\beta_0, \theta_1, -\beta_1, \theta_{2b}, -\alpha, \theta_3], r_l = [1, 1, z_{lb}, x_{lb}, q_{lb}, Lq_{lb}, k_l]^T$ and $r_b = [1, 1, z_{lb}, x_{lb}, q_{lb}, Bq_{lb}, k_b]^T$. From this system it is easy to see that

$$E[p_{lb}|s_b = 1, s_l = 1] = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + E[\epsilon_{lb}|s_b = 1, s_l = 1],$$
(16)

where $E[\epsilon_{lb}|s_b = 1, s_l = 1]$ may be different from zero, generating the selectivity bias. Here and in the next section we focus on the rate equation, derivations for the quantity equation follow consequently. The model is close to selection models proposed in the labour market literature, see Ham (1982), Poirer (1980) and Dahl (2002) among the others.¹³ While in labour market models usually agents are split in two sets (workers and firms) and have only

 $^{^{13}}$ A notable example of an empirical study using selection models in monetary economics is Fecht et al. (2011). A remarkable application of a double selection model in the labor literature is Accetturo and Infante (2013).

one match per time, in OTC markets agents can have multiple simultaneous matches (trades) with everybody.

Observe that if we have a panel and the unobservables do not vary (i) across time observations and (ii) pairs of banks, we can also use bank fixed effects to control for the endogeneous selection process. Nevertheless, the approach proposed here is preferable because it is effective even if conditions (i) and (ii) do not hold, which is possible in money markets.

At this point, it is worth to note that, if bilateral searching and monitoring efforts are not observable by the econometrician (that is usually the case) and ϵ_{lb} is correlated with them -i.e. with $s_{b,l}$, $s_{l,b}$ or $m_{l,b}$, it implies that $\sigma_{\epsilon v_B} \neq 0$, $\sigma_{\epsilon v_L} \neq 0$ and $\sigma_{v_L v_B} \neq 0$. It is also possible that such costs are correlated with some of the observables characteristics included in the regression. This fact could impair OLS estimates of equation (6) because of selection on observables bias. Note that it could be the case if we want to estimate the effect of bank's size or nationality on rates and they are correlated with search and monitor activities. For instance, if there are monitoring or searching economies of scale the effect of banks' size can be biased. In addition, if searching costs vary by countries, nationality dummies may be biased as well. In Appendix A we describe three simple examples where selectivity bias could matter, the estimators used in the empirical analysis are detailed in Appendix B.

As we observe data on bilateral trades, a matching model is a possible framework for studying the equilibrium formation of these relationships.¹⁴ In this literature, researchers use different methods to estimate matching model models (Fox, 2009). See Chiappori and Salanié (2016) and Graham (2011) for surveys on the econometrics of matching models. Among the others, Boyd et al. (2013) use the simulated method of moments, Ackerberg and Botticini (2002) and Akkus (2008) employ likelihood methods, Fox (2008) introduce a maximum score estimator, Dagsvik (2000) and Choo and Siow (2006) study games with logit methods, Sørensen (2007) explores the use of a matching model to parametrically selection correct an auxiliary outcome equation. As we want to model the outcome of a trade, but the outcome is only observed in the data for realized matches, we use the same approach of Sørensen (2007).

6 Empirical Analysis

Our final goal is to study the features of the unsecured money market for euro funds during the European sovereign debt crisis. In this section we apply the proposed dyadic econometric model to a unique dataset containing banks characteristics and bilateral trades to study how banks nationality and balance sheet composition affected rates, quantities and the probability of bilaterally trading.

6.1 Data

To answer our research question, we need three fundamental types of information: (i) bilateral trades, (ii) banks characteristics and (iii) banking groups structure.

Interbank bilateral trades are the main ingredient that we need, then information about counterparties allows us to model trading outcomes as a function of observable characteristics. Likewise knowing whether banks belong to the same group is fundamental in order to get unbiased estimates. Intragroup trades follow completely different logics, consequently the

¹⁴A matching model takes a set of payoffs or outputs for all possible matches and produces a set of matches where no couple would prefer to deviate and become matched, instead of their assigned matches.

same pair of banks, in terms of observable covariates, may be associated with unrealistic rates. Often intragroup rates are linked to extreme (close to the ceiling or to the floor) or average values that are totally inelastic to macro conditions or banks credit risk.

Unfortunately, for that period there is no institutional dataset containing such information. In this section we detail how we constructed the dataset.

To te best of our knowledge, this is the first study to collect and jointly analyze the information from transaction-level data and characteristics of global banking groups operating in euro.

Bilateral Trades. We need information about euro interbank money markets during the sovereign debt crisis. For that period, the most complete information available is from the application of the Furfine algorithm to TARGET2 (T2) data.¹⁵ ¹⁶

Alternative information on euro money market transactions could be found in: (i) reporting by the major banks in the euro area on their overnight lending rates and volumes (which make up the EONIA panel); (ii) data on individual exchanges on the Italian electronic trading platform e-MID; (iii) data on individual trades on the Spanish domestic market MID; and (iv) data on domestic and cross-border lending and borrowing for Greek banks. All of these datasets are not preferable to application of the Furfine algorithm to T2 data. EONIA panel data only refer to the aggregated daily overnight transactions of the major money-market actors in the euro area. e-MID data account for less than 20 percent of overall interbank transactions in the euro area and are, especially since mid-2011, mainly representative of Italian banks. Similarly, MID and Greek data mainly reflect the Spanish and Greek interbank markets.¹⁷

The majority of the interbank transactions are settled in T2, it allows banks to settle large value payments on their accounts in central bank money. The reserve requirement is managed on these accounts, so participating banks have to exchange money in T2 to meet the reserve requirement and make other payments. Furfine (1999) proposed an algorithm that matches the loan and its repayment, both settled on the RTGS payment system, identifying the market microstructure. Furfine's algorithm is used to detect loans from a set of payments. By definition a loan consists of two payments, the first equal to l and the second equal to l(1+i), where i is the interest rate. The algorithm matches those two legs, see Furfine (1999) for details. See Armantier and Copeland (2012) for an assessment of the quality of Furfine-based algorithms on Fedwire data and Rempel (2016) for a refinement on the Canadian payment system. Arciero et al. (2016) applied this criterion to payments settled in T2, augmenting the maturity spectrum by up to one year and making several refinements to the algorithm. Rainone and Vacirca (2016) extended the algorithm when rates are zero or negative. Arciero et al. (2016) contains detailed information about the algorithm and its practical implementation in T2 for the period under analysis. Importantly, the Eurosystem implementation enhance the algorithm to reduce the uncertainty of the results. Moreover, the results have been validated against two external data sources: (i) individual EONIA panel contributions and (ii) e-MID

¹⁵Recently the Eurosystem started a project to get transaction-by-transaction data from a sample of EU reporting agents covering the secured, unsecured, foreign exchange swap and euro overnight index swap money market segments (called MMSR, money market statistical reporting). See https://www.ecb.europa.eu/stats/financial_markets_and_interest_rates/money_market/html/index.en.html.

¹⁶T2 is the European RTGS (Real Time Gross Settlement) payment system. For more information about TARGET2 see http://www.ecb.europa.eu/paym/t2/html/index.en.html.

 $^{^{17}}$ See Arciero et al. (2016) for more detail.

transaction-level data. The validation shows that the algorithm's performance is considerably reassuring, particularly in the overnight segment.¹⁸ This is the reason why in this paper we focus only on this segment. In Appendix C we describe the potential issues of dealing with data from the Furfine algorithm and the necessary assumptions to make in order to get consistent estimates from the dyadic model outlined in Section 5.

Banking Groups Structure. To get unbiased estimates we need to correctly exclude intragroup loans. Such trades do not follow market rules and thus their inclusion generates distortion. Again, the challenge here is to get the most complete information available. Given that market participants are worldwide, there is no institutional database mapping the structure of these multinational banks.

Nevertheless, banks connect to payment systems using networks that enable them to send and receive financial transactions. SWIFT (Society for Worldwide Interbank Financial Telecommunication) is the provider of secure financial messaging services for T2 during the period under analysis. Consequently, banks operating in euro have to access such interbank network directly or indirectly. SWIFT provides users with a Bank Directory containing all the reference data to prepare, validate and process payments to other banks. This directory contains also the banking group structure. We exploit such source of information and exclude intragroup loans using the multinational group structure derived from the directory.¹⁹

Banks Characteristics. The most important observable banks characteristics that should influence trading outcomes are probably nationality and balance sheet composition. The first, is particularly important in the euro zone, as a currency union. Markets were still fragmented and sovereign risk was extremely heterogeneous during the period under analysis, then having market data with participants from many different countries is a key ingredient of our analysis. To exploit such a feature we need to precisely measure this characteristic and include it in the model. Another important information to include is the balance sheet size and composition. At the time of the sovereign debt crisis there were no official consolidated information collected at the Eurosystem level, and, given the presence of non European banks in the market, we would have been forced to search for additional data anyway. We thus have to rely on available information from public databases of banks worldwide. Our balance sheet data are taken from Bankscope.²⁰ An important advantage of this annual balance sheet dataset is that it allows us to control for a broader set of bank-specific variables. In addition, everybody can get this data, so lenders can also monitor their borrowers using such data source. Let us now detail the variables that we included in the model.

Total assets expressed in millions of euros captures the dimension of each bank. Balance sheet items are included as percentages of total assets. On the asset side Loans, Fixed Assets and Non-Earning Assets are included. Other Earning Assets are dropped because of collinearity. On the liability side, Deposits and Short-term Funding, Other Interest Bearing Liabilities, Other Reserves and Equity are included. Loan Loss Reserves and Other (Non-Interest Bearing) are dropped. Banks operating in the system are not necessarily from eurozone countries,

¹⁸This result is in sharp contrast with the recent paper by Armantier and Copeland (2012) assessing the quality of Furfine's algorithm implemented at the Federal Reserve Bank of New York against a data set of bilateral transactions between two large U.S. dealers.

¹⁹For this purpose the field Parent BIC code is considered to consolidate the group of accounts.

²⁰The construction of this dataset was done in cooperation with Giovanni di Iasio, Marco Rocco and Francesco Vacirca.

even though the majority of market participants are from countries whose central bank is part of the Eurosystem. Country dummies are included for: Italy, France, Spain, Netherlands, Greece, Ireland, United Kingdom, Austria, Portugal, Luxembourg, Cyprus, Switzerland, Finland and Belgium. Germany is the reference category. Other European countries are grouped in one dummy as well as the US, Japan and other non-European countries. We consolidated banking groups according to the SWIFT Bank Directory, as outlined above. The nationality is thus taken from the head of the group. Descriptives statistics for three maintenance periods and variables detailed description are provided in Table 1.

6.2 Empirical Specification

To ease the computational burden we assume that θ_{2b} and γ_{2l} are equal to zero in this section.²¹ Given the wide time span (from may 2008 to the end of 2012), the data is aggregated at the maintenance period level,²² then we repeatedly estimate the parameters in equation (18) for each time interval. Observe that $b_{0,t}$ thus captures systematic and macroeconomic factors affecting time interval t. In the empirical application we use the following set of information. W.r.t. the variables defined in Section 5, we set $x_{bl_{2}t} = [B_{l,t}, C_{l,t}, B_{b,t}, C_{b,t}, g_{lb,t-1}], z_{bl,t} =$ $[B_{l,t}, C_{l,t}, B_{b,t}, C_{b,t}], k_{b,t} = [\bar{p}^B_{b,t-1}, q^B_{b,t-1}, n^B_{b,t-1}], k_{l,t} = [\bar{p}^L_{l,t-1}, q^L_{l,t-1}, n^L_{l,t-1}].$ $B_{i,t}$ and $C_{i,t}$ contain respectively the information about the balance sheet structure and nationality of bank *i* at time t. $g_{ij,t}$ is equal to 1 if a loan with i as borrower and j as lender was observed at time t, it basically captures the persistence in the relationship between i and j, which may play a role in determine the rate of a loan (Affinito, 2012; Cocco et al., 2009). $\bar{p}_{i,t}^B$ and $\bar{p}_{i,t}^L$ are the average rates experienced respectively as borrower and as lender at time t by bank i, while $q_{i,t}^B$ and $q_{i,t}^L$ are the values exchanged respectively as borrower and as lender at time t by the bank i. $n_{i,t}^B$ and $n_{i,t}^L$ are the number of counterparties respectively as borrower and as lender at time t by the bank i. These last three variables can be powerful explanatory variables respectively for borrower and lender shadow rates and work as exclusion restrictions in the estimation process.²³ The presence of many financial crises during the time span considered provides frequent exogenous shocks to banks' shadow rates. For example, many lenders left the market suddenly. In our framework, it translates into considerable changes of the supply acceptable region (the blue areas in Figure 3) and the consequent exclusion of these banks from the market, no matter who the possible counterparts are. In the empirical section presented below, the robustness of such specification is also tested rather than directly imposed on the data.

6.3 Main Results

The description is organized as follows. Firstly, we focus on two maintenance periods to describe in detail the outcome of the empirical model and the estimation procedure. The

 $^{^{21}}$ In other words, we assume that the quantity slope is not borrower(lender)-specific. It is not a very restrictive assumption in this context, because it just implies that constraints in (9) and (10) only impose absolute upper (lower) bounds that are borrower(lender)-specific but not sensitive to the loan's quantity.

²²The maintenance period is the reference time interval (roughly four or six weeks long) during which the amount of central bank money is averaged on the reserve accounts. It makes this time interval the best choice to aggregate data. Quantity are summed over the time interval, rates are averaged.

 $^{^{23}}$ In the case of collinearity problems (Leung and Yu, 1996), it follows that the strength of our estimation approach depends on the extent to which these variables impact on the selection process but not in the bilateral price formation.

main aim is to check whether shadow rates exist and can bias some parameter estimates, we thus compare the estimates obtained with and without controlling for the selectivity bias. Secondly, a time series analysis is used to describe the evolution of trading patterns over time. In the first part, tables are used to describe the results, while graphical tools are needed to track time evolution in the second part. The baseline results are referred to the parametric estimation when not specified.

6.3.1 Evidence of Shadow Rates

To describe in detail the estimation procedure, we focus on two maintenance periods going from 2010-01-20 to 2010-02-09 (MP1) and from 2009-02-11 to 2009-03-10 (MP2). The aim of this analysis is twofold. First, we want to provide a consistent characterization of the probability, the rate and the quantity exchanged through bilateral trades in the OTC market for euro funds. Second, we are interested in assessing the existence of shadow rates, to do that we compare the proposed econometric model with a standard dyadic regression where we do not take into account any endogeneity issue.

The results for the quantities are presented in Table 2 and 3, while those for the rates are represented in Table 5 and 4. The first two columns report the estimates from a simple dyadic regression, the second two estimates using our methodology, the last two report the T-stat difference and its p-value. In Appendix D we report the first steps, showing the results for the likelihood to trade as a borrower or lender -i.e. the selection equations-. This is the information that we use to control for the selectivity bias.

For quantities, Table 2 and 3 show that the selection correction terms (the Mills ratios) are significantly different from zero for both MP1 and MP2. Controlling for the selection mechanism considerably impacts on the coefficient of French borrowers in MP1 (Table 2) but not in MP2 (Table 3). In terms of economic implications, our methodology indicates that French banks borrowed on average 35 millions more than German banks, while no significant difference is detected by a standard dyadic regression. Furthermore, the effect of the amount of money previously borrowed switches from positive to a more rational negative sign.

Considering the rate function, also in Table 4 and 5 we can see that the parameters of lender and borrower Mills ratios are significantly different from zero. In Table 4 we witness substantial changes in the effects of the size of the borrower, which is underestimated by one third and more importantly the effects of the share of loans in the asset composition, which shifts from not significant to negatively impacting the loan's rate. Indeed, in Table 5, if we focus on the significant coefficients, we can notice that there are differences between the estimates with an without the selection correction. As an example, big banks are able to earn more as lenders, while they save more when acting as borrowers, taking advantage of a higher bargaining power, in line with the evidences found in the literature (see Angelini et al., 2011, among others). A simple regression underestimates the first effect and overestimates the second one, producing biased evidences.²⁴ In terms of estimated profitability, the average net interest margin is downward biased by almost 10% if selection bias is not taken into account.

As a whole, it seems that endogeneity does not strongly impact all the coefficients after controlling for the wide set variables included in this study. Nevertheless, some parameters are

²⁴Another notable difference is between the coefficient of Greek, Portuguese and Cypriot borrowers, they are systematically overestimated by a simple regression. This difference points at tighter shadow rates when the borrower is from these countries. The selection bias comes from borrowers more prone to pay a higher rate. Such additional information would not be available without the proposed method.

not consistently estimated during some time periods. In the time series analysis that follows, we analyze more systematically and comprehensively the factors determining rates, quantities and trading link formation, always controlling for such potential bias.

6.3.2 Trading Patterns during the Sovereign Crisis

Here we want to study in detail the estimated coefficients and their variation during a long period which is strongly characterized by financial instability and uncertainty. The main aim is to understand how nationality and balance sheet composition influenced bilateral outcomes over time. With our econometric framework we can shed some light on the evolution of the aggregate time series presented in Section 3, and understand what are the banks characteristics that mainly drove such macro dynamics, such as the remarkable increase in rates dispersion and skewness depicted in panels (e) and (f) of Figure 2. Let us start with the likelihood to trade as a borrower or lender, Figures 4 - 6 describe these results. We then move to rates and quantities, presented in Figures 7 - 16. The results provide an unbiased characterization of the probability, the rate and the quantity exchanged of bilateral trades in the unsecured money market for euro funds from may 2008 to the end of 2012.

Trading Probability. Let us start with the characterization of the probability to bilaterally trade. Here we concentrate on the most interesting results, the rest can be found in Appendix E.

From Figure 4 we can see that the lender's balance sheet structure does not show a negative or positive persistent effect on the probability of trading, only the size matters (*Total assets*). Bigger banks are more likely to trade, which is consistent with a core-periphery structure (Craig and Von Peter, 2014; intVeld and van Lelyveld, 2014).

On the other hand, the borrower's balance sheet composition does matter in determining such probability and shows important variation through the time. Indeed, the borrower generates the risk behind the loan and then the probability of that trade. In particular, Figure 4 witnesses an increasing importance of *Equity* over the time span considered. The marginal effect of the weight of equity on the total assets of the borrower almost doubled (moving from 0.4 to 0.8). An inverse pattern is showed by the weight of *Loans* on total assets. The magnitude of the negative marginal effect steadily increased between the first and second peak of the crisis, signaling a higher difficulty of being financed by banks more exposed to illiquid assets. Both these evidences highlight the increasing selection of sound borrowers into the market by more worried lenders over time, and thus an active monitoring by the latter. After the LTROs, such selection of high-equity more-liquid borrowers disappeared, possibly because of the full allotment provided by the lender of last resort -i.e. the Eurosystem- (Garcia-de Andoain et al., 2016).²⁵ Appendix E reports the additional results on the effects of nationality.

Figure 5 and 6 show respectively the effect of lender and borrower's previous activity on the same side of the market. The number of past counterparties (in the right panel) positively affects the probability of trading for both the lender and the borrower.²⁶

 $^{^{25}}$ Remarkably, borrowers with higher short-term funding and non-earning assets increased their presence in the market over time. On the other hand, an higher share of fixed assets provoked a lower probability of borrowing.

²⁶Exchanged quantities (in the middle panel) have a more ambiguous effect, showing negative and positive effects, depending on the time the loan is agreed. Past rates (in the left panel) have more frequently a negative effect, this is especially true for the borrower.

Rates and quantities. Let us move to the second steps, the rate and quantity functions. From Figure 7 and 8 we can see that the coefficients of the Mills ratios are sometimes significantly different from zero, signaling that the selectivity bias can be an issue in several time periods.

For what concerns rates (Figure 9 - 11), country dummies show the most interesting evidences. Indeed they approximate borrower's counterparty risk at the country level.²⁷ Greek borrowers paid systematically higher rates after the subprime crisis and after the first phase of the sovereign crisis to then almost disappear from the market. Portuguese borrowers show a systematic positive spread in the period under analysis with an increasing trend, which stopped only after the LTROs in late 2011. Cypriot borrowers, when able to access the market, paid the highest interest rates especially after the subprime crisis. Italian and Spanish borrowers experienced an increasing spread from the first phase of the sovereign crisis towards the second. On the other side of the market, it is also interesting to notice that after the second phase of the crisis, lenders from some of the most stressed countries, namely Italy, Spain and Greece, extremely increased their rates, because of a remarkable increase in payment shocks during this period. More specifically, huge net outflows of central bank money occurred, as witnessed by the increase of TARGET2 balances (Figure 12). Most of these payments were related to securities trading reflecting the portfolio choices of investors (see Beck et al., 2016). In response to this higher uncertainty about payments, banks responded by becoming more reluctant to lend excess reserves when reserves were high and by becoming more aggressive in bidding for borrowed reserves when balances were low, a mechanism close to what happened during the 2007-08 financial crisis for the fed funds market (see Ashcraft et al., 2011). After the first LTRO such spreads were cleared from the market by the huge amount of liquidity provided by the Eurosystem. The balance sheet composition effects are less strong than nationality ones. Nevertheless, there are several periods in which balance sheet composition matters in determining the rate. Most interestingly, only bank size seems to have a systematic impact. Big banks seem to charge higher interest rates as lenders and pay lower rates as borrowers, that is coherent with bigger banks playing as intermediaries in the market.

Quantity time series also show interesting results. From Figure 13 we notice negative coefficients with U-shapes between the first and second peaks of the crisis for lender country dummies.

A significant negative coefficient here means that on average banks from that country lent less than the reference (German banks), which is coherent with a liquidity hoarding mechanism (see Acharya and Merrouche, 2012; Acharya and Skeie, 2011; Afonso et al., 2011; Heider et al., 2015, among others). From this figure it seems that banks started to hoard liquidity around the first peak of the crisis with a different timing and intensity depending on the nationality. With the second phase kicking in, it seems that all the banks started to hoard, thus clearing such country-time-level heterogeneity. In the same time interval we witness a inverse U-shape for Italian and French borrowers. It means that in this time interval Italian and French banks were not only lending less money but were also borrowing more money.²⁸

²⁷Country dummies can capture also redenomination risk.

²⁸Balance sheet and borrower's country effects are reported in Appendix E.

6.4 Diagnostics

Mills Ratios Linearity. An issue that may arise when using the Mills ratios is that they can be linear functions of other covariates included in both the outcome equations and the first steps. Table 6 shows the explanatory power that the controls used in both rate and quantity equations have on the borrower and lender Mills ratios. Even though some regressors show a significant correlation with the ratios, overall the unexplained component is relevant as witnessed by the difference between the \bar{R}^2 and one. The lower panels of Figure 14 report the time series of the \bar{R}^2 computed after having regressed the two Mills ratios on the other regressors, both are almost always remarkably far from 1 and with different values, signaling that the linear dependence is not a big issue over this time span. Nevertheless, for the first four time periods the Mills ratios are perfectly explained by the other regressors (Figure 14). In addition to the Mills ratios coefficients (Figure 7 and 8), the issue affects only the first four estimates of the time-variant constant and their standard errors (see Figure 15), highlighting that the value assumed by the ratios is almost constant among the units for these time periods. This is because the Mills ratio is linear for some intervals of its arguments (see Leung and Yu (1996) and Puhani (2000)).

Functional Assumptions. Normality was assumed throughout the previous section. To test assumption's correctness, we use the semiparametric estimator outlined in Appendix B. The semiparametric method is able to capture non linear relationships w.r.t. a parametric estimator and does not depend on the Mills ratio's functional form. Table 7 and 8 compare the coefficients estimated using both the parametric and semiparametric methods during the MP1. On average they are very close, not highlighting a prominent departure from the normality assumption, thus the relative figures are not reported for the sake of brevity. Nevertheless, it is suggested to compute both these estimators to check this assumption and see whether some parameters are badly estimated under the distributional assumptions imposed to the data. Furthermore, the first four estimates of the time-variant constant and their standard errors are no longer badly computed, as shown in Figure 16 (comparing to panel (c) of Figure 15). This highlights the importance of considering both a parametric and semiparametric approach when selectivity issues are taken into account.

Exclusion Restrictions. As hinted in Section 6.2, the choice of variables that play as exclusion restrictions is fundamental to robustly identify the outcome equation's parameters. If these variables have an impact on the outcome and are correlated with some regressors at the same time, the correction terms may just capture this feature. It would imply that the Mills ratios only correct for the omission of observable variables included in the first step (shadow rates) but not included in the outcome equations (rates and quantities). If so, the inclusion of the correction terms would just be fictitiously informative. To check for such an issue, it is possible to test whether the inclusion of the Mills ratios changes the correlation between the residuals from the outcome equations and the exclusion restrictions. Table 9 reports the results of two regressions, with and without correction terms, for both rate and quantity equation and a test for a significant difference between the two. For all the exclusion restrictions there is no significant difference between the coefficients.

7 Concluding Remarks

In this paper we studied pairwise trading in the unsecured interbank market for euro funds during the European sovereign debt crisis. The goal of our analysis was to understand how banks characteristics affected the probability of trading, bilateral rates and quantities.

To embed the specific features of the OTC trading in the unsecured money market, we proposed a dyadic econometric model with shadow rates to simultaneously study trading probability, rates and quantities. In doing so, we discuss the potential bias emerging when the counterparties endogenously select each other into bilateral trades, for example when monitoring and searching efforts are endogenous. We propose a simple characterization of this counterparty selection bias as a specification error and present a consistent estimation methodology.

We built and used a unique dataset containing the characteristics of banks operating worldwide and bilateral trades in the unsecured interbank market for euro funds. We first found mild evidence regarding the existence of shadow rates after controlling for the wide set of controls included in this study, we then used our consistent estimator to study trade patterns during the European debt crisis.

As regards the formation of trading links, borrower's balance sheet composition matters in determining such probability and shows important variation through the time. We witness an increasing importance of equity over the time span considered. The marginal effect of the weight of equity on the total liabilities of the borrower almost doubled between the two peaks of the crisis. An inverse pattern is showed by the weight of loans on total assets. The magnitude of the negative marginal effect steadily increased between the first and second peak of the crisis, signaling a higher difficulty of being financed by banks more exposed to illiquid assets. Both evidences highlight the increasing selection of sound borrowers into the market by more worried lenders over time, and thus an active monitoring by the latter. After the implementation of the LTROs by the Eurosystem, such selection of high-equity and moreliquid borrowers disappeared, possibly because of the full allotment provided by the lender of last resort.

We find a substantial dispersion in rates and quantities driven by banks nationality and balance sheet, especially during the peak of the crisis, shedding light on new aspects featuring the unsecured money market for euro funds. Before the Eurosystem LTROs, we found that high market fragmentation and rate dispersion were mostly driven by borrowers characteristics, while liquidity rationing was largely explained by lenders characteristics. Among the many new evidences collected, we showed how borrower balance sheet composition and nationality impacted dramatically on the probability of borrowing money in general and especially at low rates, which is coherent with a credit-risk story and an active monitoring by lenders. Furthermore, we witnessed a differential liquidity hoarding activity across space and time between the two phases of the sovereign crisis mainly explained by lenders nationality. More Specifically, Italian and Spanish borrowers paid an increasing spread from the first phase of the sovereign crisis through the second one. On the other side of the market, it was also interesting to notice that after the second phase of the sovereign crisis, lenders from some of the most stressed countries, namely Italy, Spain and Greece, extremely increased their rates, because of the sudden market stress and the scarcity of liquidity providers.

References

- Accetturo, A. and Infante, L. (2013). Skills or culture? an analysis of the decision to work by immigrant women in italy, *IZA Journal of Migration* 2(1): 1–21.
- Acharya, V. V. and Merrouche, O. (2012). Precautionary hoarding of liquidity and interbank markets: Evidence from the subprime crisis, *Review of Finance* p. rfs022.
- Acharya, V. V. and Skeie, D. (2011). A model of liquidity hoarding and term premia in inter-bank markets, *Journal of Monetary Economics* 58(5): 436–447.
- Ackerberg, D. A. and Botticini, M. (2002). Endogenous matching and the empirical determinants of contract form, *Journal of Political Economy* 110(3): 564–591.
- Affinito, M. (2012). Do interbank customer relationships exist? and how did they function in the crisis? learning from italy, *Journal of Banking & Finance* **36**(12): 3163–3184.
- Afonso, G., Kovner, A. and Schoar, A. (2011). Stressed, not frozen: The federal funds market in the financial crisis, *The Journal of Finance* 66(4): 1109–1139.
- Afonso, G. and Lagos, R. (2015). Trade dynamics in the market for federal funds, *Econometrica: Journal of the Econometric Society* pp. 263–313.
- Akkus, O. (2008). Endogenous matching and the determinants of underpricing in ipo markets.
- Angelini, P., Nobili, A. and Picillo, C. (2011). The interbank market after august 2007: what has changed, and why?, *Journal of Money, Credit and Banking* **43**(5): 923–958.
- Arciero, L., Heijmans, R., Heuver, R., Massarenti, M., Picillo, C. and Vacirca, F. (2016). How to measure the unsecured money market? the eurosystem's implementation and validation using target2 data, *International Journal of Central Banking* 2: 247–280.
- Armantier, O. and Copeland, A. (2012). Assessing the quality of furfine-based algorithms, Federal Reserve Bank of New York staff report 575.
- Ashcraft, A., McAndrews, J. and Skeie, D. (2011). Precautionary reserves and the interbank market, *Journal of Money, Credit and Banking* **43**(s2): 311–348.
- Babus, A. and Hu, T.-W. (2017). Endogenous intermediation in over-the-counter markets, Journal of Financial Economics.
- Bech, M. and Monnet, C. (2016). A search-based model of the interbank money market and monetary policy implementation, *Journal of Economic Theory* **164**: 32–67.
- Beck, R., Georgiadis, G. and Gräb, J. (2016). The geography of the great rebalancing in euro area bond markets during the sovereign debt crisis, *Journal of Empirical Finance*.
- Blasques, F., Bräuning, F. and Van Lelyveld, I. (2016). A dynamic network model of the unsecured interbank lending market.
- Boyd, D., Lankford, H., Loeb, S. and Wyckoff, J. (2013). Analyzing the determinants of the matching of public school teachers to jobs: Disentangling the preferences of teachers and employers, *Journal of Labor Economics* **31**(1): 83–117.

- Caballero, R. J. and Krishnamurthy, A. (2008). Collective risk management in a flight to quality episode, *The Journal of Finance* **63**(5): 2195–2230.
- Cameron, A. C. and Miller, D. L. (2014). Robust inference for dyadic data, *Technical report*, Working Paper.
- Chiappori, P.-A. and Salanié, B. (2016). The econometrics of matching models, *Journal of Economic Literature* **54**(3): 832–861.
- Choo, E. and Siow, A. (2006). Who marries whom and why, *Journal of political Economy* **114**(1): 175–201.
- Cocco, J. F., Gomes, F. J. and Martins, N. C. (2009). Lending relationships in the interbank market, *Journal of Financial Intermediation* **18**(1): 24–48.
- Cosslett, S. R. (1983). Distribution-free maximum likelihood estimator of the binary choice model, *Econometrica* pp. 765–782.
- Craig, B. and Von Peter, G. (2014). Interbank tiering and money center banks, Journal of Financial Intermediation 23(3): 322–347.
- Dagsvik, J. K. (2000). Aggregation in matching markets, *International Economic Review* **41**(1): 27–58.
- Dahl, G. B. (2002). Mobility and the return to education: Testing a roy model with multiple markets, *Econometrica* **70**(6): 2367–2420.
- Duffie, D., Gârleanu, N. and Pedersen, L. H. (2005). Over-the-counter markets, *Econometrica* 73(6): 1815–1847.
- Fecht, F., Nyborg, K. G. and Rocholl, J. (2011). The price of liquidity: The effects of market conditions and bank characteristics, *Journal of Financial Economics* **102**(2): 344–362.
- Flannery, M. J. (1996). Financial crises, payment system problems, and discount window lending, *Journal of money, credit and banking* pp. 804–824.
- Fox, J. T. (2008). Estimating matching games with transfers, *Technical report*, National Bureau of Economic Research.
- Fox, J. T. (2009). Structural empirical work using matching models, *New Palgrave Dictionary* of *Economics*. Online edition.
- Frutos, J. C., Garcia-de Andoain, C., Heider, F. and Papsdorf, P. (2016). Stressed interbank markets: evidence from the european financial and sovereign debt crisis.
- Furfine, C. (1999). The microstructure of the federal funds market, Financial Markets, Institutions & Instruments 8(5): 24–44.
- Furfine, C. (2001). Banks as monitors of other banks: Evidence from the overnight federal funds market, *The Journal of Business* 74(1): 33–57.

- Gabrieli, S. and Georg, C.-P. (2014). A network view on interbank market freezes, Available at SSRN 2542162.
- Garcia-de Andoain, C., Heider, F., Hoerova, M. and Manganelli, S. (2016). Lending-of-last-resort is as lending-of-last-resort does: Central bank liquidity provision and interbank market functioning in the euro area, *Journal of Financial Intermediation*.
- Garcia-de Andoain, C., Hoffmann, P. and Manganelli, S. (2014). Fragmentation in the euro overnight unsecured money market, *Economics Letters* **125**(2): 298–302.
- Gaspar, V., Quirós, G. P. and Mendizábal, H. R. (2008). Interest rate dispersion and volatility in the market for daily funds, *European Economic Review* **52**(3): 413–440.
- Gourieroux, C., Monfort, A., Renault, E. and Trognon, A. (1987). Generalised residuals, Journal of Econometrics **34**(1): 5–32.
- Graham, B. (2011). Econometric methods for the analysis of assignment problems in the presence of complementarity and social spillovers, *Handbook of social economics* 1: 965– 1052.
- Ham, J. C. (1982). Estimation of a labour supply model with censoring due to unemployment and underemployment, *The Review of Economic Studies* **49**(3): 335–354.
- Heckman, J. (1974). Shadow prices, market wages, and labor supply, *Econometrica* **42**(4): 679–694.
- Heckman, J. (1979). Sample selection bias as a specification error, *Econometrica* 47(1): 153–61.
- Heckman, J. J. and Robb, R. (1985). Alternative methods for evaluating the impact of interventions: An overview, *Journal of Econometrics* **30**(1): 239–267.
- Heider, F., Hoerova, M. and Holthausen, C. (2015). Liquidity hoarding and interbank market rates: The role of counterparty risk, *Journal of Financial Economics* **118**(2): 336–354.
- Ichimura, H. (1993). Semiparametric least squares (sls) and weighted sls estimation of singleindex models, *Journal of Econometrics* 58(1): 71–120.
- intVeld, D. and van Lelyveld, I. (2014). Finding the core: Network structure in interbank markets, *Journal of Banking & Finance* **49**: 27–40.
- Kenny, D. A., Kashy, D. A. and Cook, W. L. (2006). Dyadic data analysis, Guilford Press.
- Khan, S. (2013). Distribution free estimation of heteroskedastic binary response models using probit/logit criterion functions, *Journal of Econometrics* **172**(1): 168–182.
- Klein, R. W. and Spady, R. H. (1993). An efficient semiparametric estimator for binary response models, *Econometrica* pp. 387–421.
- Knight, F. H. (2012). Risk, uncertainty and profit, Courier Dover Publications.

- Leung, S. F. and Yu, S. (1996). On the choice between sample selection and two-part models, Journal of econometrics 72(1): 197–229.
- Manski, C. F. (1975). Maximum score estimation of the stochastic utility model of choice, Journal of econometrics **3**(3): 205–228.
- Mayordomo, S., Abascal, M., Alonso, T. and Rodriguez-Moreno, M. (2015). Fragmentation in the european interbank market: Measures, determinants, and policy solutions, *Journal* of Financial Stability 16: 1–12.
- Newey, W. K. (2009). Two-step series estimation of sample selection models, *The Econometrics Journal* **12**(s1): S217–S229.
- Plackett, R. L. (1945). A reduction formula for normal multivariate integrals, *Biometrika* 41(3/4): 351–360.
- Poirer, D. (1980). Partial observability in bivariate probit models, *Journal of Econometrics* 12: 209–217.
- Powell, J. L., Stock, J. H. and Stoker, T. M. (1989). Semiparametric estimation of index coefficients, *Econometrica: Journal of the Econometric Society* pp. 1403–1430.
- Puhani, P. (2000). The heckman correction for sample selection and its critique, *Journal of economic surveys* 14(1): 53–68.
- Rainone, E. (2015). Testing information diffusion in the decentralized unsecured market for euro funds, *Bank of Italy Temi di Discussione (Working Paper) No* **1022**.
- Rainone, E. and Vacirca, F. (2016). Estimating money market microstructure with negative and zero interest rates, *Bank of Italy Temi di Discussione (Working Paper) No* **1059**.
- Rempel, M. (2016). Improving overnight loan identification in payments systems, Journal of Money, Credit and Banking 48(2-3): 549–564.
- Sørensen, M. (2007). How smart is smart money? a two-sided matching model of venture capital, *The Journal of Finance* **62**(6): 2725–2762.

Tables

Table 1: Observed loans descriptives

Maintenance peri	od	200	09-03-11	- 2009-0	4-07	201	10-11-10 -	- 2010-1	2-07	20	11-09-14	- 2011-1	0-11
Variable Loan	Description	mean	std	min	max	mean	std	min	max	mean	std	min	max
Rate	Interest rate paid	0.83	0.20	0.21	2.50	0.30	0.07	0.12	1.15	0.55	0.18	0.15	1.70
Quantity	Quantity exchanged (mil- lions)	16.19	53.42	0.05	1033.16	16.06	45.13	0.07	664.29	19.50	98.50	0.05	3138.16
Lender	Loops expressed as percent												
A loan	ages of lender total assets	0.57	0.20	0.00	0.90	0.59	0.20	0.00	0.89	0.58	0.20	0.00	0.91
A fix as	Fixed assets expressed as per- centages of lender total assets	0.01	0.01	0.00	0.14	0.01	0.01	0.00	0.09	0.01	0.01	0.00	0.09
A non ern	as percentages of lender total assets	0.07	0.07	0.00	0.96	0.07	0.06	0.00	0.96	0.07	0.08	0.00	0.96
L dep sh fun	Deposits and short-term fund- ing expressed as percentages of lender total assets	0.62	0.17	0.00	0.99	0.62	0.16	0.00	0.98	0.62	0.17	0.00	0.98
L oth int bea	Other interest bearing liabili- ties expressed as percentages	0.25	0.17	0.00	0.87	0.25	0.15	0.00	0.85	0.25	0.16	0.00	0.85
L oth res	Other reserves expressed as percentages of lender total as-	0.01	0.01	0.00	0.13	0.01	0.00	0.00	0.04	0.01	0.01	0.00	0.20
L equ	sets Equity expressed as percent-	0.08	0.04	0.00	0.60	0.08	0.05	0.00	0.56	0.08	0.04	0.00	0.56
A tot proof	ages of lender total assets Total assets expressed in mil-	10.00	2.22	2.06	14.54	0.02	0.00	2 50	14 51	10.04	2.26	2.60	14 51
A tot asset	lions of euros Dummy variable taking value	10.00	2.22	3.00	14.04	9.92	2.33	5.59	14.01	10.04	2.30	3.09	14.01
IT	equal to 1 if the lender is from this country (or set of coun- tries) and zero otherwise.	0.44	0.50	0.00	1.00	0.54	0.50	0.00	1.00	0.47	0.50	0.00	1.00
FR	""	0.05	0.21	0.00	1.00	0.04	0.21	0.00	1.00	0.04	0.20	0.00	1.00
ES NL	""	0.05	0.22	0.00	1.00	$0.04 \\ 0.02$	0.19	0.00	1.00	0.05	0.21	0.00	1.00
GR	""	0.03	0.16	0.00	1.00	0.03	0.17	0.00	1.00	0.04	0.19	0.00	1.00
IE	27 27 27 27	0.02	0.13	0.00	1.00	0.01	0.08	0.00	1.00	0.00	0.07	0.00	1.00
US/JAP/EX	""	0.02	0.15	0.00	1.00	0.01	0.11	0.00	1.00	0.02	0.14	0.00	1.00
AT	** **	0.06	0.24	0.00	1.00	0.07	0.26	0.00	1.00	0.07	0.25	0.00	1.00
PT	22.22	0.04	0.19	0.00	1.00	0.03	0.17	0.00	1.00	0.04	0.19	0.00	1.00
LU	22.22	0.01	0.11	0.00	1.00	0.00	0.07	0.00	1.00	0.02	0.13	0.00	1.00
CH	37 33	0.01	0.11	0.00	1.00	0.01	0.07	0.00	1.00	0.00	0.07	0.00	1.00
FI	** **	0.00	0.06	0.00	1.00	0.00	0.04	0.00	1.00	0.00	0.04	0.00	1.00
EUEX	»» »»	0.08	0.27	0.00	1.00	0.06	0.25	0.00	1.00	0.07	0.26	0.00	1.00
BE Borrower	32.32	0.00	0.06	0.00	1.00	0.00	0.00	0.00	0.00	0.00	0.06	0.00	1.00
A loan	Loans expressed as percent- ages of borrower total assets	0.57	0.19	0.00	0.91	0.60	0.19	0.00	0.87	0.57	0.20	0.00	0.87
A fix as	Fixed assets expressed as per- centages of borrower total as-	0.01	0.02	0.00	0.09	0.01	0.01	0.00	0.09	0.01	0.02	0.00	0.09
A non ern	Non -earning assets expressed as percentages of borrower to-	0.07	0.07	0.00	0.96	0.07	0.06	0.00	0.96	0.06	0.06	0.00	0.96
L dep sh fun	Deposits and short-term fund- ing expressed as percentages	0.57	0.17	0.00	0.93	0.59	0.16	0.00	0.94	0.60	0.15	0.00	0.94
L oth int bea	of borrower total assets Otherinterest bearing liabili- ties expressed as percentages	0.31	0.16	0.00	0.95	0.28	0.16	0.00	0.92	0.28	0.15	0.00	0.92
L oth res	of borrower total assets Other reservers expressed as percentages of borrower total	0.01	0.01	0.00	0.08	0.01	0.00	0.00	0.03	0.01	0.00	0.00	0.08
Lequ	assets Equity expressed as percent-	0.07	0.03	0.00	0.29	0.07	0.03	0.00	0.27	0.07	0.03	0.00	0.27
A tot asset	ages of borrower total assets Total assets expressed in mil-	11.28	1.97	3.06	14.54	10.74	1.98	3.99	14.51	10.94	2.15	5.23	14.51
11 000 00000	lions of euros Dummy variable taking value equal to 1 if the borrower is	11.20	1101	0.00	11101	10111	1100	0.00	11101	10101	2.10	0.20	11101
IT	from this country (or set of countries) and zero otherwise.	0.42	0.49	0.00	1.00	0.55	0.50	0.00	1.00	0.47	0.50	0.00	1.00
FR	""	0.06	0.23	0.00	1.00	0.03	0.16	0.00	1.00	0.08	0.27	0.00	1.00
ES	22.22	0.04	0.20	0.00	1.00	0.04	0.19	0.00	1.00	0.05	0.21	0.00	1.00
GR	""	0.01	0.09	0.00	1.00	0.01 0.02	0.11	0.00	1.00	0.01	0.12	0.00	1.00
IE	33 37	0.02	0.13	0.00	1.00	0.01	0.09	0.00	1.00	0.00	0.07	0.00	1.00
UK	»» »»	0.03	0.17	0.00	1.00	0.02	0.15	0.00	1.00	0.02	0.13	0.00	1.00
US/JAP/EX	»»»	0.05	0.21	0.00	1.00	0.04	0.19	0.00	1.00	0.04	0.19	0.00	1.00
AT PT	33 33	0.06	0.23	0.00	1.00	0.06	0.23	0.00	1.00	0.05	0.21	0.00	1.00
LU	""	0.02	0.13	0.00	1.00	0.02	0.15	0.00	1.00	0.02	0.13	0.00	1.00
CY	»» »»	0.01	0.11	0.00	1.00	0.01	0.12	0.00	1.00	0.01	0.10	0.00	1.00
CH	»» »»	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
FI	27 27 27 27	0.00	0.04	0.00	1.00	0.00	0.05	0.00	1.00	0.00	0.00	0.00	0.00
BE		0.06	$0.23 \\ 0.07$	0.00 0.00	1.00	0.03	0.18	0.00 0.00	1.00	0.03	$0.17 \\ 0.05$	0.00	1.00
Pairs observed			14	134			16	13			13	391	

Notes: Three representative maintenance periods are described. Other maintenance periods descriptives are not reported for the sake of brevity and are available upon request. Fixed assets are also known as tangible assets or property, plant, and equipment, they are illiquid assets and cannot easily be converted into cash. See Bankscope website for a more detailed description of the balance sheet data collection.

Dependent Variable: bilateral quantity exchanged							
	Simple r	egression	Selection	correction	T-:	stat difference	
Mills Ratio Borrower			-115.01	199*** 951)			
Mills Ratio Lender			-85.51	74 ***			
			(17.8	3718)			
	Lender	Borrower	Lender	Borrower	Lender	Borrower	
A loan	6.3986	-6.1036	-0.8143	-14.1645 *	0.6512	0.6644	
	(7.8825)	(8.6316)	(7.7816)	(8.5254)	[0.7425]	0.7467]	
A fix as	-205.9081	-214.8(43)	-144.4210	-27.0155	-0.2707	-0.7261	
A non ern	66 8535 **	(105.0041)	(139.2440) 73.2480 ***	(102.3413) -72 5851 **	[0.3933] _0 1600	$\begin{bmatrix} 0.2340 \end{bmatrix}$ 0.1076	
A non em	(28,5485)	(32.7724)	(27.9638)	(32.0641)	$\begin{bmatrix} 0.4365 \end{bmatrix}$	[0.5428]	
L dep sh fun	34.8874	-44.4600	18.3525	-19.3109	0.4476	-0.6039	
F	(26.3047)	(29.4818)	(25.9386)	(29.4155)	[0.6727]	[0.2730]	
L oth int bea	34.4045	-37.5091	30.1779	-23.8505	0.1122	-0.3289	
	(26.9349)	(29.5910)	(26.3464)	(29.1361)	[0.5446]	[0.3711]	
L oth res	-94.2805	-464.5184	-26.2109	-478.2933	-0.1601	0.0284	
	(303.8844)	(346.9461)	(297.3783)	(339.3746)	[0.4364]	[0.5113]	
L equ	26.2915	55.8195	15.5751	42.4104	0.1941	0.1410	
	(39.4601)	(67.9548)	(38.6153)	(66.5232)	[0.5769]	[0.5561]	
A tot asset	0.8051	0.2353	-1.7011 *	-1.8950 *	1.9349	1.3480	
	(0.8898)	(1.1078)	(0.9412)	(1.1271)	[0.9734]	[0.9110]	
IT	-5.3730	2.1513	-0.8136	10.9651 *	-0.5891	-0.9788	
	(5.5120)	(6.3591)	(5.4327)	(6.3753)	[0.2780]	[0.1640]	
\mathbf{FR}	6.4208	9.6956	-7.8743	35.6712 ***	1.2724	-1.8280	
	(7.8721)	(9.4197)	(8.0154)	(10.6393)	[0.8982]	[0.0339]	
ES	7.3184	-9.9276	8.7138	-9.1562	-0.1092	-0.0625	
	(9.1304)	(8.8161)	(8.9429)	(8.6224)	[0.4565]	[0.4751]	
NL	0.1365	-20.6351	-7.6851	-15.2179	0.5706	-0.3066	
CD	(9.7475)	(12.6118)	(9.6375)	(12.3784)	[0.7158]	[0.3796]	
GR	10.7974	-10.2980	10.9168	-5.7408	-0.0050	-0.1782	
	(17.1002)	(18.2703)	(10.7803)	(17.8894)	0.4980]	[0.4293]	
UK	9.0100	-0.0704	1.9690	-11.9164	0.4023	0.2000	
US/LAP/EX	(11.0014)	9.2517	-0.5150	14 4030	-0.0103	_0.2810	
05/5/1/1/1/	(10.5874)	(13 3213)	(10.3500)	(13.0572)	[0.4023]	[0 3894]	
АT	-2 7670	-5 0445	(10.3530) 1 7754	-0.4491	-0 5499	_0.4910	
111	(5.8758)	(6.6712)	(5,8060)	(6,5636)	[0 2913]	[0.3118]	
РТ	5.3247	-22.1638 **	7.1816	-14.7469	-0.1513	-0.5494	
	(8.7659)	(9.5934)	(8.5896)	(9.4998)	[0.4399]	[0.2914]	
CY	()	-27.2089	()	-18.2724	[]	-0.3817	
		(16.7152)		(16.3943)		[0.3514]	
EUEX	-0.9160	-22.6624 ***	-1.4285	-16.1408 **	0.0535	-0.5929	
	(6.8343)	(7.8358)	(6.7206)	(7.7185)	[0.5213]	[0.2767]	
Rates at t-1	-2797.4123 *	1946.0673	-2922.3237 *	733.2042	0.0557	0.5280	
	(1601.9363)	(1630.5859)	(1566.9273)	(1618.2008)	[0.5222]	[0.7012]	
Value exchanged at t-1	0.0546 ***	0.0203 ***	0.0395 ***	-0.0261 **	2.5184	3.5207	
	(0.0037)	(0.0066)	(0.0048)	(0.0114)	[0.9940]	[0.9998]	
Number of counterparts at t-1	1.4814	-1.8501	0.9372	-1.2246	0.3077	-0.3441	
	(1.2597)	(1.2955)	(1.2412)	(1.2749)	[0.6208]	[0.3654]	
Connection at t-1	13.462	27 ***	12.996	65 ***		0.1194	
_	(2.7)	910)	(2.73)	323)		[0.5475]	
Constant	10.0	226	167.23	67 ***		-2.5259	
50	(41.4	498)	(46.4)	1306)		[0.0058]	
R^2	0.3	402	0.3	692			
Time interval			2010-01-20) - 2010-02-09			
Maturity			1 to	3 days			
Observations				1067			

Notes: *: p < 0.10; **: p < 0.05; **: p < 0.01. Standard errors are reported in round brackets, p-values in squared brackets. Estimates are from the linear model described in Section 5, the empirical specification is described in Section 6.2. Controls are described in Table 1, their effect are reported for borrowers and lenders in the respective columns. In the selection correction model the Mills ratios are added. "Number of counterparts at t-1" is the number of trading counterparties over the previous maintenance period as lender or borrower. "Value exchanged at t-1" is the average value of loans over the previous maintenance period as lender or borrower. "Rates at t-1" is the average rate over the previous maintenance period as lender or borrower. "Connection at t-1" is equal to 1 if the two banks exchanged money in the previous maintenance period. The T-stat differences are computed between the same coefficient estimated by simple regression and the selection correction procedure (H_0 : equivalence of the two coefficients). Only country fixed effects with more than 1% of observations are included in the model. Estimates are from the parametric model.

Dependent Variable: bilateral quanti	ity exchanged					
	Simple re	gression	Selection of	correction	T-stat o	lifference
Mills Ratio Borrower			-5.50)61 867)		
Mills Ratio Lender			-16.9 (158.2	521 2507)		
	Lender	Borrower	Lender	Borrower	Lender	Borrower
A loan	1.7606	6.6400	1.5393	5.2504	0.0117	0.0427
A fix as	(13.3207) -149.1683	(10.2507) -432.8531	(15.3544) -147.1083	(28.2352) -417.7091	-0.0058	-0.0384
A non ern	(252.6273) -94.0567 **	(276.8203) -27.6420	(252.9713) -93.3866 **	(280.8647) -21.9008	[0.4977] -0.0108	[0.4847] -0.0720
L dep sh fun	(43.7994) -32.7732	(54.8992) 7.4151	(43.8824) -32.2845	(57.7941) 10.0088	[0.4957] -0.0086	[0.4713] -0.0356
L oth int bea	(40.3783) -44.4192	$(51.2354) \\ -9.5320$	(40.4386) -43.7578	$(51.8562) \\ -7.2354$	[0.4966] -0.0109	[0.4858] -0.0323
L oth res	(43.0131) -15.0961	(50.0215) -14.1784	(43.0941) -16.8335	(50.5855) -37.5391	$[0.4957] \\ 0.0041$	$[0.4871] \\ 0.0324]$
L equ	(303.1681) -157.6289 **	(507.6146) 108.3875	(303.4820) -156.8065 **	(513.0739) 105.2249	[0.5016]	$[\begin{array}{c} 0.5129 \\ 0.0196 \end{array}]$
A tot asset	(67.6506) -3.1419 **	$(113.8246) \\ 0.3531$	(67.7542) -3.2202 **	$(114.3171) \\ 0.0744$	$[\begin{array}{c} 0.4966 \\ 0.0385 \end{array}]$	$[\begin{array}{c} 0.5078 \\ 0.0992 \end{array}]$
IT	(1.3967) -3.6031	(1.6454) 18.9753 **	$(1.4797) \\ -3.3973$	(2.2764) 19.6287 **	$[0.5154] \\ -0.0171$	$[0.5395] \\ -0.0480$
FR	(8.4757) 19.7853 *	(9.4222) 31.1235 **	(8.5394) 19.8211 *	(9.8294) 30.4662 **	[0.4932] -0.0022	$[\begin{array}{c} 0.4809 \\ 0.0378 \end{array}]$
ES	$(11.7449) \\ 2.2977$	$(12.1966) \\ 4.7209$	(11.7730) 2.2235	$(12.3621) \\ 4.7715$	$[0.4991] \\ 0.0043$	[0.5151] -0.0025
NL	$(12.3112) \\ 6.5555$	(13.9928) 5.5763	$(12.3322) \\ 6.8656$	$(14.1062) \\ 5.7060$	[0.5017] -0.0142	[0.4990] -0.0043
GR	(15.4091) - 14.4963	(21.4893) 1.7322	(15.4726) -14.4807	(21.6248) 1.5081	[0.4943] -0.0006	$[0.4983] \\ 0.0096$
UK	(17.4432) -21.6804	(16.5155) 14.8554	(17.4585) -21.4928	(16.6484) 13.7790	[0.4997] -0.0095	$[\begin{array}{c} 0.5038 \\ 0.0524 \end{array}]$
US/JAP/EX	(13.9617) 5.5621	$(14.3478) \\ -7.4766$	(14.0123) 5.5527	$(14.7128) \\ -6.3888$	$[0.4962] \\ 0.0004$	[0.5209] -0.0354
AT	$(17.5499) \\ 13.3601$	(21.5685) 1.1660	(17.5802) 13.2064	$(21.8964) \\ 0.6481$	$[\begin{array}{c} 0.5002 \\ 0.0074 \end{array}]$	$[\begin{array}{c} 0.4859 \\ 0.0346 \end{array}]$
РТ	$(14.6754) \\ 0.2844$	(10.5086) -2.6162	$(14.7053) \\ 0.3742$	$(10.6555) \\ -1.9588$	$[0.5030] \\ -0.0065$	[0.5138] -0.0236
СҮ	(9.7857) 3.5884	$(19.6470) \\ 1.0048$	$(9.8091)\ 3.5018$	$(19.7927) \\ 0.3234$	$[\begin{array}{c} 0.4974 \\ 0.0036 \end{array}]$	$[\begin{array}{c} 0.4906 \\ 0.0238 \end{array}]$
EUEX	(16.9875) 3.1793	(20.0984) 9.5185	$(17.0083) \\ 3.3796$	$(20.3666) \\ 9.4619$	$0.5014 \\ -0.0159$	$[0.5095] \\ 0.0038$
Rates at t-1	(8.8992) 1322.1041	(10.4184) -716.7974	(8.9523) 1311.6763	$(10.5524) \\ -747.8046$	$[\begin{array}{c} 0.4937 \\ 0.0062 \end{array}]$	$[\begin{array}{c} 0.5015 \\ 0.0320 \end{array}]$
Value exchanged at t-1	(1180.6889) 0.0858 ***	(680.7336) 0.0440 ***	$(1182.0920) \\ 0.0857 ***$	(689.9159) 0.0438 ***	$[\begin{array}{c} 0.5025 \\ 0.0108 \end{array}]$	[0.5128] 0.0168
Number of counterparts at t-1	(0.0084) -5.8225	(0.0074) 1.8141	(0.0084) -5.7800	(0.0075) 1.7943	$[0.5043] \\ -0.0076$	$[\begin{array}{c} 0.5067 \\ 0.0061 \end{array}]$
Connection at t-1	(3.9305) 18.832	(2.2935) 4 ***	$\begin{array}{r} (3.9358) \\ 18.7308 \\ *** \end{array} (2.2977)$		[0.4970] 0.0	[0.5024] 168
Constant	(4.25) 63.3 (71.3)	594) 398 843)	(4.2764) 79.0490 (153.6449)		$\begin{bmatrix} 0.5067 \\ -0.0927 \\ 0.4631 \end{bmatrix}$	
$ar{R}^2$	0.19	020	0.19)33		
Time interval Maturity			2009-02-11 - 20 1 to 3 da	009-03-10 ays		
Observations			1183	-		

Table 3: Quantity equation MP2

Notes: See 2.

	Simple r	egression	Selection	correction	T-stat	difference
Mills Ratio Borrower			0.03	98** 163)		
Mills Ratio Lender			0.047 (0.0	′5*** 182)		
	Lender	Borrower	Lender	Borrower	Lender	Borrower
A loan	0.0114 (0.0209)	-0.0181 (0.0243)	0.0100 (0.0209)	-0.0936 ** (0.0411)	0.0484 [0.5193]	1.5798 [0.9428]
A fix as	-0.0528 (0.4118)	0.4215 (0.4083)	-0.0167 (0.4111)	0.2554 (0.4145)	-0.0620 [0.4753]	0.2854 [0.6123]
A non ern	(0.0342) (0.0706)	0.2607 *** (0.0878)	(0.0374) (0.0705)	0.2745 *** (0.0879)	-0.0323	-0.1105
dep sh fun	0.0595 (0.0658)	0.1794 ** (0.0795)	0.0624 (0.0657)	0.1629 ** (0.0796)	-0.0307	0.1462
L oth int bea	0.0017 (0.0702)	0.2194 *** (0.0802)	0.0031 (0.0701)	0.2098 *** (0.0802)	-0.0141 [0.4944]	0.0847
L oth res	0.7474 (0.4916)	0.8611 (0.8198)	0.7566 (0.4906)	0.5835 (0.8348)	-0.0133 [0.4947]	0.2373
Lequ	0.0154 (0.1083)	-0.0662 (0.1849)	0.0203 (0.1081)	-0.0591 (0.1848)	-0.0315 [0.4875]	-0.0270 [0.4892]
A tot asset	0.0139 *** (0.0021)	-0.0205 *** (0.0024)	0.0123 *** (0.0022)	-0.0271 *** (0.0034)	0.5373 [0.7044]	1.5839 [0.9433]
Т	-0.0020 (0.0132)	0.0263 * (0.0146)	-0.0001 (0.0131)	0.0298 ** (0.0148)	-0.1041 [0.4586]	-0.1684 [0.4331]
FR	-0.0622 *** (0.0190)	0.0181 (0.0199)	-0.0603 *** (0.0190)	0.0207 (0.0199)	-0.0721 [0.4713]	-0.0926 [0.4631]
ES	0.0189 (0.0198)	0.0429 * (0.0226)	0.0200 (0.0198)	0.0530 ** (0.0228)	-0.0386 [0.4846]	-0.3139 [0.3768]
NL	0.0426 * (0.0253)	0.2690 *** (0.0349)	0.0466 * (0.0253)	0.2793 *** (0.0350)	-0.1113 [0.4557]	-0.2072 [0.4179]
GR	0.0298 (0.0286)	$\begin{array}{c} 0.1107 \ ^{***} \\ (0.0268) \end{array}$	0.0279 (0.0285)	0.1203 *** (0.0270)	0.0466 [0.5186]	-0.2512 [0.4008]
JK	-0.0114 (0.0229)	0.0311 (0.0233)	-0.0090 (0.0229)	0.0330 (0.0233)	-0.0731 [0.4709]	-0.0564 [0.4775]
JS/JAP/EX	-0.0676 ** (0.0286)	$0.0733 ** \\ (0.0338)$	-0.0642 ** (0.0286)	0.0779 ** (0.0338)	-0.0847 [0.4663]	-0.0965 [0.4616]
АT	-0.0519 ** (0.0236)	0.0349 ** (0.0170)	-0.0508 ** (0.0236)	0.0394 ** (0.0171)	-0.0313 [0.4875]	-0.1860 [0.4262]
PT	-0.0222 (0.0158)	$\begin{array}{c} 0.1046 \ ^{***} \\ (0.0320) \end{array}$	-0.0214 (0.0157)	$\begin{array}{c} 0.1129 \ ^{***} \\ (0.0322) \end{array}$	-0.0398 0.4841	-0.1813 [0.4281]
CY	-0.0149 (0.0278)	$\begin{array}{c} 0.2731 \ ^{***} \\ (0.0326) \end{array}$	-0.0159 (0.0277)	$\begin{array}{c} 0.2807 \ ^{***} \\ (0.0328) \end{array}$	0.0250 [0.5100]	-0.1649 [0.4345]
EUEX	-0.0614 *** (0.0143)	$\begin{array}{c} 0.0912 \ ^{***} \\ (0.0169) \end{array}$	-0.0580 *** (0.0143)	$0.0979 *** \\ (0.0171)$	-0.1703 [0.4324]	-0.2788 [0.3902]
Constant	0.977 (0.1	79 *** 112)	$1.492 \\ (0.2)$	6 *** 261)	-2 [0.	.0431 0206]
Connection at t-1	0.0 (0.0	082 064)	0.0 (0.0)	049 066)	0. [0.	3656 6426]
Quantity exchanged	-0.0 (0.0	0000 000)	-0.0 (0.0	000 000)	0. [0.	0433 5173]
\bar{R}^2	0.3	028	0.3	572		
Fime interval Maturity Observations			2009-02-11 1 to 3	- 2009-03-10 3 days 183		

Table 4: Rate equation MP2

Notes: See 2.

=

	Simple 1	egression	Selection	correction	T-stat	difference
Mills Ratio Borrower			0.03	398** 0163)		
Mills Ratio Lender			0.04 (0.0	75***)182)		
	Lender	Borrower	Lender	Borrower	Lender	Borrower
A loan	0.0163 (0.0133)	-0.0460 *** (0.0145)	0.0145 (0.0136)	-0.0396 *** (0.0149)	0.0841	-0.0298
A fix as	(0.2910) (0.2739)	(0.9290 ***) (0.2981)	(0.3530) (0.2753)	(0.3061)	-0.2312 [0.4086]	-0.4014
A non ern	0.1006 ** (0.0488)	(0.0532) (0.0554)	0.1114 ** (0.0491)	-0.0619 (0.0570)	0.0468	-0.0175
dep sh fun	0.1067 ** (0.0444)	(0.0345) (0.0487)	(0.0101) 0.1316^{***} (0.0452)	(0.0327) (0.0526)	-0.2102	-0.0476
oth int bea	0.0461 (0.0458)	(0.0573) (0.0491)	0.0689 (0.0467)	0.0556 (0.0535)	-0.0939 [0.4626]	-0.1052
L oth res	-0.0826 (0.5045)	-0.4602 (0.5828)	0.1963 (0.5165)	-0.6389 (0.5985)	-0.1107	0.1622 [0.5644]
Lequ	0.0258 (0.0667)	0.3316 *** (0.1155)	0.0457 (0.0677)	0.3033 ** (0.1177)	-0.1446 [0.4425]	0.0104
A tot asset	0.0066 ***	-0.0052 *** (0.0018)	0.0080 *** (0.0016)	-0.0047 ** (0.0021)	-1.2526 [0.1053]	-1.1767
Т	0.0096 (0.0089)	(0.0011) (0.0094)	0.0162 * (0.0094)	-0.0032 (0.0096)	-0.3541 [0.3617]	0.7159 [0.7629]
FR	-0.0029 (0.0131)	-0.0038 (0.0157)	-0.0017 (0.0141)	(0.0020) (0.0219)	-0.0252	0.2393
ES	(0.0131) (0.0138) (0.0156)	-0.0061	(0.0111) (0.0159) (0.0157)	(0.0210) -0.0054 (0.0152)	0.2196	0.2542
NL	-0.0128	0.0061 (0.0219)	(0.0101) -0.0124 (0.0170)	(0.0102) 0.0038 (0.0221)	-0.1127	0.4059
GR	-0.0411	(0.0215) 0.0784 ** (0.0315)	(0.0110) -0.0352 (0.0297)	(0.0221) 0.0749 ** (0.0316)	-0.0169	0.2719
UK	(0.0290) -0.0079 (0.0190)	(0.0313) 0.0079 (0.0160)	(0.0297) -0.0226 (0.0212)	(0.0310) 0.0039 (0.0165)	-0.0523	$\begin{bmatrix} 0.0071 \\ 0.2362 \\ 0.5033 \end{bmatrix}$
US/JAP/EX	(0.0133) -0.0311 * (0.0170)	-0.0294	(0.0212) -0.0345 *	-0.0345	0.2097	$\begin{bmatrix} 0.3933 \end{bmatrix}$ 0.4740
AT	(0.0179) -0.0089 (0.0000)	(0.0250) -0.0150 (0.0112)	(0.0183) -0.0064 (0.0100)	(0.0232) -0.0138 (0.0114)	0.0799	0.5175
PT	(0.0099) 0.0310 ** (0.0145)	0.0566 ***	(0.0100) 0.0360 **	(0.0114) 0.0561 ***	-0.0056	0.4380
CY	(0.0145)	(0.0166) 0.1003 ***	(0.0146)	(0.0167) 0.0973 ***	[0.4978]	$\begin{bmatrix} 0.6693 \\ 0.3012 \end{bmatrix}$
EUEX	-0.0104	0.0110	-0.0114	(0.0289) 0.0094 (0.0149)	0.1273	
Constant	(0.0116)	(0.0135)	-0.0	(0.0142) 0734	[0.5506]	[0.6386] .0777
Connection at t-1	-0.0	081 *	-0.0	076 *	[0 -(0.6167
Quantity exchanged	(0.0 (0.0 (0.0	045) 0000 0001)	(0.0 -0.0 (0.0	0046) 0000 0001)	0] -([0	0.2688] 0.7851 0.2163]
\bar{R}^2	0.2	2080	0.2	2172		
Fime interval Maturity Observations			2010-01-2 1 to	0 - 2010-02-09 5 3 days 1067		

Table 5: Rate equation MP1

Notes: See 2.

=

Dependent Variable:						
	Borrower 1	Mills Ratio	Lender	Mills Ratio		
	Lender	Borrower	Lender	Borrower		
A loan	-0.0283	0.0753 ***	0.0570 **	-0.0759 ***		
	(0.0253)	(0.0275)	(0.0227)	(0.0246)		
A fix as	0.0790	-4.8675 ***	-1.9494 ***	0.5688		
	(0.5199)	(0.5498)	(0.4661)	(0.4930)		
A non ern	0.1833 **	0.1091	-0.0853	-0.1204		
	(0.0932)	(0.1060)	(0.0836)	(0.0950)		
L dep sh fun	-0.0233	-0.0123	-0.2581 ***	-0.0590		
	(0.0844)	(0.0933)	(0.0757)	(0.0836)		
L oth int bea	-0.0212	-0.0935	-0.1106	-0.0759		
	(0.0876)	(0.0940)	(0.0785)	(0.0843)		
L oth res	0.7661	5.4197 ***	-2.3072 ***	-1.7341 *		
	(0.9622)	(1.1015)	(0.8627)	(0.9877)		
L equ	-0.1447	-0.0873	-0.1664	0.1092		
	(0.1274)	(0.2211)	(0.1142)	(0.1982)		
A tot asset	-0.0128 ***	-0.0605 ***	-0.0419 ***	-0.0092 ***		
	(0.0025)	(0.0030)	(0.0022)	(0.0027)		
IT	-0.0156	0.1700 ***	-0.0800 ***	0.0526 ***		
FD	(0.0167)	(0.0170)	(0.0150)	(0.0152)		
FR	0.0218	0.1204 ***	-0.0281	0.0107		
70	(0.0251)	(0.0298)	(0.0225)	(0.0267)		
ES	0.0469	0.1888 ***	0.0629 **	-0.0452 *		
NT	(0.0297)	(0.0282)	(0.0267)	(0.0253)		
NL	0.0246	0.3494 ***	-0.0769 ***	-0.0319		
CD	(0.0320)	(0.0404)	(0.0287)	(0.0363)		
GR	-0.0012	0.2253^{***}	-0.0140	0.0664		
111/	(0.0567)	(0.0599)	(0.0509)	(0.0537)		
UK	(0.0475)	$0.1(83^{+++})$	-0.0708	-0.0377		
	(0.0379)	(0.0300)	(0.0340)	(0.0269)		
US/JAP/EX	-0.0080	0.3487	0.1185	0.0287		
	(0.0341)	(0.0427)	(0.0305)	(0.0383)		
AI	-0.0240	(0.0207)	$(0.0437)^{+++}$	(0.0044)		
DT	(0.0169)	(0.0207)	(0.0109)	(0.0100)		
r I	-0.0362	(0.0208)	(0.0290)	(0.0021)		
CV	(0.0277)	(0.0308)	(0.0248)	(0.0270)		
01		(0.2392)		(0.0378)		
FUFY	0.0208	0.1758 ***	0.0101	(0.0430)		
LUEA	(0.0293)	(0.0253)	(0.0191)	(0.0004)		
Connection at $t = 1$	-0.049)2 ***	-0.0	(0.0221)		
Connection at $t = 1$	-0.042	084)	-0.0-	100 0075)		
Constant	1 133	0 ***	1 33	870 ***		
Constant	(0.1	991)	1.50	1158)		
	(0.1		(0.			
\bar{B}^2	0.6	559	Ω	4865		
Time interval	9010_01_90 - 9010_09_00					
Maturity	2010-01-20 - 2010-02-09 1 to 3 days					
Observations		1	1067			
2.3501.4010110						

Table 6: Diagnostics - Mills Ratio collinearity

Notes: See 2. Borrower and Lender Mills ratios are regressed on the controls listed to check potential collinearity of the ratios with the regressors.

Dependent Variable: bila	iteral rate				
	Para	metric	Semi	parametric	
	Lender	Borrower	Lender	Borrower	
A loan	0.0163	-0.0460 ***	0.0145	-0.0396 ***	
	(0.0133)	(0.0145)	(0.0136)	(0.0149)	
A fix as	0.2910	0.9290 ***	0.3530	1.0132 ***	
	(0.2739)	(0.2981)	(0.2753)	(0.3061)	
A non ern	0.1006 **	-0.0532	0.1114 **	-0.0619	
	(0.0488)	(0.0554)	(0.0491)	(0.0570)	
L dep sh fun	0.1067 **	0.0345	0.1316 ***	0.0327	
	(0.0444)	(0.0487)	(0.0452)	(0.0526)	
L oth int bea	0.0461	0.0573	0.0689	0.0556	
	(0.0458)	(0.0491)	(0.0467)	(0.0535)	
L oth res	-0.0826	-0.4602	0.1963	-0.6389	
_	(0.5045)	(0.5828)	(0.5165)	(0.5985)	
L equ	0.0258	0.3316 ***	0.0457	0.3033 **	
	(0.0667)	(0.1155)	(0.0677)	(0.1177)	
A tot asset	0.0066 ***	-0.0052 ***	0.0080 ***	-0.0047 **	
	(0.0015)	(0.0018)	(0.0016)	(0.0021)	
IT	0.0096	0.0011	0.0162 *	-0.0032	
	(0.0089)	(0.0094)	(0.0094)	(0.0096)	
\mathbf{FR}	-0.0029	-0.0038	-0.0017	0.0020	
	(0.0131)	(0.0157)	(0.0141)	(0.0219)	
ES	0.0138	-0.0061	0.0159	-0.0054	
	(0.0156)	(0.0150)	(0.0157)	(0.0152)	
NL	-0.0128	0.0061	-0.0124	0.0038	
	(0.0168)	(0.0219)	(0.0170)	(0.0221)	
GR	-0.0411	0.0784 **	-0.0352	0.0749 **	
	(0.0296)	(0.0315)	(0.0297)	(0.0316)	
UK	-0.0079	0.0079	-0.0226	0.0039	
	(0.0199)	(0.0160)	(0.0212)	(0.0165)	
US/JAP/EX	-0.0311 *	-0.0294	-0.0345 *	-0.0345	
	(0.0179)	(0.0230)	(0.0183)	(0.0232)	
AT	-0.0089	-0.0150	-0.0064	-0.0138	
	(0.0099)	(0.0113)	(0.0100)	(0.0114)	
PT	0.0310 **	0.0566 ***	0.0360 **	0.0561 ***	
	(0.0145)	(0.0166)	(0.0146)	(0.0167)	
CY		0.1003 ***		0.0973 ***	
		(0.0289)		(0.0289)	
EUEX	-0.0104	0.0110	-0.0114	0.0094	
	(0.0116)	(0.0135)	(0.0119)	(0.0142)	
Connection at $t = 1$	-0.0	081 *	-(0076 *	
$\frac{1}{2}$	-0.0 (n r	0045)	-(0.0046)	
Quantity exchanged	(0.0	0000	(0.00407	
gaanning chemangeu	(n r	0001)	-	0.0001)	
Constant	0.0	089	(-0.0734	
Composito	(0.0	0744)	(0.0987)	
Time interval		2010-01-	-20 - 2010-02-09	9	
Maturity	1 to 3 days				
Observations		-	1067		
Constant Time interval Maturity Observations	0.1 (0.0	1089 [°] 10744) 2010-01- 1	-0.0734 (0.0987) -20 - 2010-02-09 to 3 days 1067		

Table 7: Diagnostics - Rate distributional assumptions

Notes: See 2. Rate is modeled using the parametric and semiparametric techniques, the coefficients of the two alternative methods are reported. Only country fixed effects with more than 1% of observations are included in the model. A power of four was used to approximate the unknown function in the semiparametric model.

	Paran	netric	Semipar	rametric		
	Lender	Borrower	Lender	Borrower		
A loan	-0.8143	-14.1645 *	4.2825	-16.6934 '		
	(7.7816)	(8.5254)	(8.0464)	(9.1175)		
A fix as	-144.4210	-27.0155	-169.1523	-22.5261		
	(159.2448)	(182.3413)	(158.1029)	(183.1893)		
A non ern	73.2480 ***	-72.5851 **	71.1222 **	-59.0340		
	(27.9638)	(32.0641)	(28.0304)	(32.1148)		
L dep sh fun	18.3525	-19.3109	27.2343	-3.3528		
	(25.9386)	(29.4155)	(26.4520)	(29.7158)		
L oth int bea	30.1779	-23.8505	42.0625	-12.4231		
	(26.3464)	(29.1361)	(26.8888)	(29.4617)		
L oth res	-26.2109	-478.2933	-49.8700	-489.819		
	(297.3783)	(339.3746)	(298.6204)	(338.6524)		
equ	15.5751	42.4104	34.9539	24.2006		
	(38.6153)	(66.5232)	(38.8946)	(66.5888)		
A tot asset	-1.7011 *	-1.8950 *	-0.9458	-2.6520 *		
	(0.9412)	(1.1271)	(1.0443)	(1.2495)		
T	-0.8136	10.9651 *	0.3116	7.7030		
	(5.4327)	(6.3753)	(5.6359)	(6.4850)		
FR	-7.8743	35.6712 ***	-7.9446	37.7282 *		
	(8.0154)	(10.6393)	(8.1462)	(11.2568)		
ES	8.7138	-9.1562	5.4231	-4.2421		
	(8.9429)	(8.6224)	(8.9677)	(8.8740)		
NL	-7.6851	-15.2179	-12.5694	-11.2910		
	(9.6375)	(12.3784)	(9.7578)	(12.3228)		
GR	10.9168	-5.7408	9.1887	-4.6172		
	(16.7803)	(17.8894)	(16.7088)	(17.7893)		
JK	1.9895	-11.9184	-7.4027	-11.0966		
/	(11.4826)	(9.3498)	(11.9258)	(9.6347)		
US/JAP/EX	-0.5150	14.4930	-4.0424	13.2878		
	(10.3590)	(13.0572)	(10.5510)	(13.0043)		
A'T'	1.7754	-0.4491	1.4700	0.0962		
	(5.8060)	(6.5636)	(5.8337)	(6.5400)		
2'1'	7.1816	-14.7469	3.9825	-10.2616		
	(8.5896)	(9.4998)	(8.6046)	(9.4808)		
JY	-1.4285	-18.2724	-2.9129	-13.8566		
	(6.7206)	(16.3943)	(6.7949)	(16.3476		
EUEX		-16.1408 **		-14.1677		
		(7.7185)		(8.0200)		
Rates at $t-1$	-2922.3237 *	733.2042	-1888.6205	318.4466		
	(1566.9273)	(1618.2008)	(1586.4385)	(1616.095		
value exchanged at $t-1$	0.0395 ***	-0.0261 **	0.0539 ***	-0.0402 *		
* * * * * * * *	(0.0048)	(0.0114)	(0.0065)	(0.0221)		
Number of counterparts at $t-1$	0.9372	-1.2246	0.2687	-0.8201		
a	(1.2412)	(1.2749)	(1.2603)	(1.2818)		
Connection at $t-1$	(2.7323)		(2.7653)			
Γime interval		2010-01-20 -	2010-02-09			
Maturity		1 to 3	days			
~. ·	1067					

Table 8: Diagnostics - Quantity distributional assumptions

Notes: See 2 and 7.

Dependent Variable: estimated residuals							
	F	Rate equation		Quantity equation			
	Simple regression	Selection correction	Δ	Simple regression	Selection correction	Δ	
Borrower rates at $t-1$	4.6453 *** (1.7471)	4.3973 ** (1.7450)	0.0710 (0.4717)	-0.0000 (1030.8095)	0.0000 (1006.9246)	-0.0000 (0.5000)	
Borrower value at $t-1$	-0.0000 *** (0.0000)	-0.0000 * (0.0000)	-0.4401 (0.3300)	0.0000 (0.0040)	0.0000 (0.0039)	(0.0000) (0.5000)	
Borrower number of counterparts at $t-1$	-0.0034 ** (0.0014)	-0.0032 ** (0.0014)	-0.0820 (0.4673)	0.0000 (0.8129)	-0.0000 (0.7940)	0.0000 (0.5000)	
Lender rates at $t-1$	$\begin{array}{c} 12.9894 \\ (2.1933) \end{array}$	$\begin{array}{c} 12.4827 & *** \\ (2.1905) \end{array}$	0.1156 (0.4540)	-0.0000 (1294.0245)	-0.0000 (1264.0406)	-0.0000 (0.5000)	
Lender value at $t-1$	-0.0000 (0.0000)	0.0000 (0.0000)	-0.2960 (0.3837)	0.0000 (0.0031)	0.0000 (0.0030)	-0.0000 (0.5000)	
Lender number of counterparts at $t-1$	-0.0107 *** (0.0017)	-0.0101 *** (0.0017)	-0.1677 (0.4334)	0.0000 (1.0158)	0.0000 (0.9923)	0.0000 (0.5000)	
Time interval Maturity Observations		:	2010-01-20 1 to 3 10	- 2010-02-09 3 days 67			

Table 9: Diagnostics - Exclusion restrictions

Notes: See 2. The residuals are from the parametric model and respectively from the rate and quantity equations for both the simple regression and the selection correction model. The residuals are regressed on the excluded variables, reported in the first column. The Δ columns report the difference between the coefficients when the correction is applied and when it is not and its standard deviation.

Figures

Figure 1: Aggregate evidence - Number of links, quantity exchanged and rate.



(c) Average rate

Notes: Each data point represents a maintenance period (the starting date is reported on the X axis). Panel (a) reports the number of trades reported in each maintenance period. Panel (b) reports the quantity exchanged in millions of euro. Panel (c) plots the weighted (by volume) average rate agreed in each period. Violet vertical line traces the first peak of the sovereign debt crisis (April 2010), the black vertical line traces the second peak of the sovereign debt crisis (August 2011), the green lines trace the dates of the two Eurosystem LTROs and the light blue line traces when the overnight deposit rate was set to zero in July 2012.



Figure 2: Market side evidence - Number of links, quantity exchanged and rate.

Notes: See Figure 1. Light shades track the interdecile range, dark shades depict the interquartile interval and the median is traced with a bold line. Panel (a) and (b) report the number of trades for each bank relative to the possible matches with other market participants, 0.01 means that a lender (borrower) trades with one borrower (lender) on 100 possible market participants. Panel (b) and (d) report the average number of millions of euro per transaction for each bank. In Panel (e) and (f) rates are averaged over the maintenance period for each market participant.



Figure 3: Lender and borrower shadow rates

Notes: Blue areas refer to lenders, red areas refer to borrowers. Panel (a) depicts two different lenders $(L_1 \text{ and } L_2)$ with different γ_0 , panel (b) represents two different lenders $(L_1 \text{ and } L_2)$ with different γ_2 , the red area refers to a borrower (B_1) . Panel (c) depicts two different borrowers $(B_1 \text{ and } B_2)$ with different θ_0 , panel (d) represents two different borrowers $(B_1 \text{ and } B_2)$ with different θ_2 , the blue area refers to a lender (L_2) .



Figure 4: Probability to trade - Lender and borrower's balance sheet effects.

Notes: See Figure 1. The X axis is the time line, each data point represents a maintenance period as in Figure 1, dates are not reported for convenience. The title of the subplot reports the name of the covariate for which the effect is measured. The bold lines represent repeated cross-section OLS estimates of a linear probability model, the dashed lines represent 95 percent confidence intervals (Probit estimates produce the same patterns and are available upon request). The dependent is a binary variable taking value equal to 1 if a trade occurred between the pair of banks. The controls (z and k) listed in Section 6.2 are on the RHS. The model is estimated including all the covariates. The bold dashed red line tracks the zero axis.

Figure 5: Probability to trade - Lender's previous activity effects.



Notes: See Figure 1 and Figure 4. r_{t-1} is the average rate during the previous maintenance period. q_{t-1} is the quantity exchanged and n_{t-1} is the number of counterparties.





Notes: See Figure 1, Figure 4 and Figure 5.

Figure 7: Rate equation - Mills ratios coefficients.



Notes: See Figure 1 and Figure 4. The black line represent the coefficient of the lender Mills ratio λ_L , the violet line represent the coefficient of the borrower Mills ratio λ_B .



Figure 8: Quantity equation - Mills ratios coefficients.

Notes: See Figure 1, Figure 4 and Figure 7.



Figure 9: Rate equation - Lender and borrower balance sheet effects.

Notes: See Figure 1 and Figure 4. Estimates are obtained using the parametric model described in the Appendix. The bold lines represent repeated cross-section OLS estimates that include the correction terms (lender and borrower Mills ratios). The controls (x) listed in Section 6.2 are on the RHS. The model is estimated including all the covariates.



Figure 10: Rate equation - Lender's country effects.

Notes: See Figure 1, Figure 4 and Figure 9. Missing observations may occur if there are no observations with that characteristic.



Figure 11: Rate equation - Borrower's country effects.

Notes: See Figure 1, Figure 4, Figure 9 and 10.



Figure 12: TARGET2 balances.

Notes: Daily TARGET2 balances expressed in billions of euro.



Figure 13: Quantity equation - Lender's country effects.

Notes: See Figure 1, Figure 4, Figure 9 and 10.

Figure 14: Rate equation diagnostics - Mills ratios non linearity and percentages of uncensored lenders and borrowers.



Notes: See Figure 1. The upper panels report the percentage of uncensored borrowers (left) and lenders (right). The lower panels report the R^2 of a regression having the borrower (left) and lender (right) Mills ratios on the LHS and all the controls in the outcome equations (x) on the RHS.





(c) Time

Notes: See Figure 1, Figure 4, Figure 9 and 10.

Figure 16: Rate equation with semiparametric estimation - Time FEs.



Notes: See Figure 1, Figure 4, Figure 9 and 10.

APPENDIX

Appendix A: Three Simple Examples with Unobservables

In this section we provide three simple examples that give insights on the endogeneity issues introduced in Section 5. In the first two, market's sides are treated separately to ease the exposition, in practice both can materialize simultaneously, further exacerbating the selectivity bias.

Endogenous Borrower Searching Costs Suppose we are interested in estimating the marginal effect β_b of a borrower exogenous dummy variable $x_b = \{0, 1\}$ on p_{lb} or q_{lb} . For example x_b takes value 1 if the bank is in country A and 0 otherwise. W.l.o.g assume that the searching costs are different from zero only for banks belonging to country A -i.e. $s_1 > s_0 = 0$ -and that ϵ_{lb} is correlated with s_b . Let us focus on rates, the same arguments apply for quantities. Such heterogeneity implies that the distribution of rates for country A borrowers is upper bounded, while for other borrowers is not. It turns out that we observe just a censored distribution of rates for country A borrowers. In the heuristic example provided in Figure A.1, this censoring downward biases the estimated difference between $E(p_{lb}|x_b = 0)$ and $E(p_{lb}|x_b = 1)$ leading it to zero instead of β_b .

Figure A.1: Borrower Correlated Unobservable Searching Costs.



Notes: Red areas refer to borrowers acceptable regions. The B_1 red area refers to borrowers in country A. The B_0 red area refers to the other borrowers. The box plots represent the distributions of data points conditional on x_b . The middle line represents the true mean while the dotted one represent the biased mean.

Endogenous Lender Monitoring Costs In a specular way we could be interested in estimating the marginal effect β_l of a lander exogenous dummy variable $x_l = \{0, 1\}$ on p_{lb} or q_{lb} . As before, assume x_l takes value 1 if the bank is in country A and 0 otherwise. Assume that the monitoring costs are different from zero for banks belonging to country A and zero for the others -i.e. $m_1 > m_0 = 0$ - and that ϵ_{lb} is correlated with m_l . This censoring implies that the distribution of rates for country A lenders is lower bounded, while for other lenders is not. In the example provided in Figure A.2, this censoring upward biases β_l leading it to zero.

Figure A.2: Lender Correlated Unobservable Monitoring Costs.



Notes: Blue areas refer to lenders. The L_1 blues area refers to lenders in country A. The L_0 blues area refers to the other lenders. The box plots represent the distributions of data points conditional on x_l . The middle line represents the true mean while the dotted one represent the biased mean.

In general, it is worth to mention that these cost-based unobservables are just two possible sources of endogeneity. Other unobservables may hamper the consistent estimation of the pairwise equations parameters in this environment. Nevertheless, the shadow rates model proposed here is general enough to control for the presence of different types of unobservable, like endogenous intermediation (see Babus and Hu, 2017, for example).

Importantly, we may conduct such an analysis to assess market fragmentation, segregation or integration. If we are interested in understanding whether borrowers from country A systemically pay more, we want a consistent estimate of β_b . Ignoring such endogeneity issue may prevent it.

Endogenous Meeting Process One possible interpretation of our empirical model is as reservation rates in a search process, where banks endogenously meet, match and exchange (Figure A.3). Upon a meeting (say l_{ij} between bank *i* and bank *j*), the rate follows according to equation (6). After seeing the rate both parties decide whether to accept the deal or not according to conditions (11). If both accept, i.e. $p_{B,j}^* \ge p_{ij} \ge p_{L,i}^*$, they trade at rate p_{ij} the quantity q_{ij} . Otherwise, they continue meeting other counterparties. Banks that do not find any deal in the market have to go to the central bank's standing facilities and lend at p_{OD} or borrow at p_{ML} . Observe that the meeting does not need to be random, because we allow for possible correlation between unobserved factors affecting the meeting probability and unobservables determining the rate and the quantity exchanged. This feature is particularly appealing for the analysis of OTC interbank markets, where many determinants of meetings, rates and quantities remain unobserved by the econometricians. Furthermore, such unobservables can be correlated among themselves and generate significant bias due to endogenous selection. In the next section we propose a method to deal with this issue and get unbiased estimates for rate and quantity parameters.

Figure A.3: Endogenous meeting process.



Notes: nodes i and j are two banks. The dotted line l_{ij} represents a meeting between i and j, the blue arrow q_{ij} represents the amount and the direction of central bank money exchanged, the blue segment on the red line marks the agreed rate of the loan, the latter is bounded by p_{OD} and p_{ML} .

Appendix B: Estimation

In this section, we propose two possible procedures to consistently estimate the parameters in the empirical model outlined in Section 5. Both apply a control function approach. The first method is parametric, while the second is semiparametric. Such an approach allows also the dyadic model to capture general equilibrium effects of reserves, payment volatility and the effects of market access by some banks.²⁹

Parametric Estimation

If we assume that f(.) in (15) is a trivariate normal, $g(p_{lb}, q_{lb})$ becomes a bivariate normal density function and $P(\cdot)$ a bivariate cumulative normal density function in (13),³⁰ so that the likelihood function is known and has nice properties. Maximizing it brings to consistent, unbiased and efficient parameter estimates. The drawback of ML estimation is that it is computational intensive, in this context we are interested in estimating these parameters for a wide time span, thus ML is excessively time demanding.³¹ Heckman (1979) proposed a two step procedure as an alternative way of estimating parameters for this kind of sample selection models. In his model one selection equation determines whether the outcome of an agent is observed or not. In our framework we have two selection equations, one for the lender and one for the borrower, thus the estimation is a little bit more complicated. Poirer (1980) investigated a similar model in which the outcome reflects the choices of two decision-makers in a different context. From this distributional assumption it thus follows that

$$E[p_{lb}|s_{b} = 1, s_{l} = 1] = \beta_{0} + \beta_{1}x_{lb} + \alpha q_{lb}$$

$$+ \frac{\sigma_{\epsilon v_{B}}}{\sigma_{v_{B}}^{2}} \frac{\phi(\kappa^{*}r_{b})\Phi((\omega^{*}r_{l} - \rho_{v_{B}v_{L}}\kappa^{*}r_{b})/(1 - \rho_{v_{B}v_{L}}^{2})^{\frac{1}{2}})}{\Phi^{2}(\kappa^{*}r_{b}, \omega^{*}r_{l}, \rho_{v_{B}v_{L}})}$$

$$+ \frac{\sigma_{\epsilon v_{L}}}{\sigma_{v_{L}}^{2}} \frac{\phi(\omega^{*}r_{l})\Phi((\kappa^{*}r_{b} - \rho_{v_{B}v_{L}}\omega^{*}r_{l})/(1 - \rho_{v_{B}v_{L}}^{2})^{\frac{1}{2}})}{\Phi^{2}(\kappa^{*}r_{b}, \omega^{*}r_{l}, \rho_{v_{B}v_{L}})},$$
(17)

²⁹Technically, it is made possible by conditioning on r_l , r_b , $u_{L,l}$ and $u_{B,b}$ that can contain these endogenous variables.

³⁰Note that given that the condition $p_{B,b}^* \ge p_{lb} \ge p_{L,l}^*$ can be represented as an interval of real numbers in \Re , $P(\cdot)$ can be computed as the difference of two univariate cumulative normal density functions.

³¹We also estimated the model using ML for one MP, results are available upon request.

where $\omega^* = \frac{1}{\sigma_{v_B}}\omega$, $\kappa^* = \frac{1}{\sigma_{v_L}}\kappa$. One way to estimate this system, is to run a bivariate probit in the first step. Using a Maximum Likelihood estimator one can simultaneously estimate the parameters of both the selection equations as well as the correlation between the two errors ($\sigma_{v_B v_L}$). Another way is to estimate separately the two selection equations, in a quasimaximum likelihood approach, ignoring the correlation between the residuals. After this step one can estimate $\sigma_{v_B v_L}$ computing the correlation between the generalized residuals (Gourieroux et al., 1987). The OLS brings to consistent parameter estimates of the following model

$$p_{lb} = \beta_0 + \beta_l x_l + \beta_b x_b + \alpha q_{lb} + \delta_B \hat{\lambda}_B + \delta_L \hat{\lambda}_L + \epsilon_{lb}$$
(18)

where $\hat{\lambda}_B$ and $\hat{\lambda}_L$ are consistent estimates of $\lambda_B = \frac{\phi(\kappa^* r_b)\Phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b)/(1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}$, $\lambda_L = \frac{\phi(\omega^* r_l)\Phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l)/(1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}$, the two multivariate Mills ratios, $\hat{\delta}_B = \frac{\hat{\sigma}_{\epsilon v_B}}{\hat{\sigma}_{v_B}^2}$ and $\delta_L = \frac{\sigma_{\epsilon v_L}}{\sigma_{v_L}^2}$. Details about how to derive and estimate the relative consistent standard errors are provided in Appendix B. Observe that a FIML method can be also used to estimate the parametric model.

Consistent Parametric Standard Errors Let us focus on the errors' conditional variance for the rate equation. Specular derivations can be done for the quantity equation, they are omitted for brevity. From system (15) and the normality assumption we have that

$$\widetilde{\sigma}_{\epsilon,bl} = E(\epsilon_i^2 | s_l = 1, s_b = 1) = \sigma_{\epsilon}^2 - \sigma_{\epsilon,v_B}^2 \kappa^* r_b \lambda_L - \sigma_{\epsilon,v_L}^2 \omega^* r_l \lambda_B$$

$$+ \nu_{lb} [2\sigma_{\epsilon,v_B} \sigma_{\epsilon,v_L} - \sigma_{v_L,v_B} (\sigma_{\epsilon,v_B}^2 + \sigma_{\epsilon,v_L}^2)] - (\sigma_{\epsilon,v_B} \lambda_L - \sigma_{\epsilon,v_L} \lambda_B)^2$$

$$= \sigma_{\epsilon,i} + \zeta_{lb},$$
(19)

where $\nu_{lb} = \phi(\kappa^* r_b, \omega^* r_l, \sigma_{v_B, v_L}) / \Phi(\kappa^* r_b, \omega^* r_l, \sigma_{v_B, v_L})$, so that the following is the estimator of σ_{ϵ} .

$$\hat{\sigma}_{\epsilon} = \frac{1}{N_u} \left(\sum_{bl \in U} d_{lb} - \hat{\zeta}_{lb} \right). \tag{20}$$

where d_{lb} are the estimated residual by OLS of model. Let us call $\tilde{\Sigma}$ the diagonal matrix containing these variances. The correct variance-covariance matrix for the estimated parameters is obtained in the following way. Given the double selection mechanism, the residuals of equation (6) are

$$e_{lb} = \delta_B(\lambda_B - \hat{\lambda}_B) + \delta_L(\lambda_L - \hat{\lambda}_L) + \epsilon_{lb}$$
⁽²¹⁾

Let $\tau = (\kappa^*, \omega^*, \rho_{v_B v_L})$ and take the fist-order approximation of $\hat{\lambda}_B$ and $\hat{\lambda}_L$

$$(\lambda_{B,i} - \hat{\lambda_{B,i}}) = \frac{\partial \lambda_{B,i}}{\partial \tau} (\tau - \hat{\tau})$$
(22)

$$(\lambda_{L,i} - \hat{\lambda_{L,i}}) = \frac{\partial \lambda_{L,i}}{\partial \tau} (\tau - \hat{\tau})$$
(23)

Let
$$X^* = (\iota, x_l, x_b, \hat{\lambda}_B, \hat{\lambda}_L), \ \beta^* = (\beta_0, \beta_l, \beta_b, \delta_B, \delta_L) \text{ and } C_i = (\delta_B \frac{\partial \lambda_{B,i}}{\partial \tau} + \delta_L \frac{\partial \lambda_{L,i}}{\partial \tau}), \text{ then}$$

$$(\hat{\beta}^* - \beta^*) = (X^{*'}X^*)^{-1}(X^{*'}e_{lb}) = (X^{*'}X^*)^{-1}(\epsilon_{lb} + C(\tau - \hat{\tau})),$$
(24)

then we have the following variances for each parameter:

$$diag(var(\hat{\beta}^*)) = (X^{*'}X^*)^{-1}X^{*'}(\tilde{\Sigma} + C var(\hat{\tau})C')X^*(X^{*'}X^*)^{-1}$$
(25)
$$= (X^{*'}X^*)^{-1}(X^{*'}\tilde{\Sigma}X^* + X^{*'}C var(\hat{\tau})C'X^*)(X^{*'}X^*)^{-1}$$

For computing this matrix we need $C = (\delta_B \frac{\partial \lambda_B}{\partial \tau} + \delta_L \frac{\partial \lambda_L}{\partial \tau})$, a $N_u \times 2k + 1$, where k is the dimension of r_b and r_l (suppose it is the same), and consequently $\frac{\partial \lambda_B}{\partial \tau}$ and $\frac{\partial \lambda_L}{\partial \tau}$. Given that

$$\lambda_B = \frac{\phi(\kappa^* r_b) \Phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b)/(1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})},$$

$$\lambda_L = \frac{\phi(\omega^* r_l) \Phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}.$$

We just need to compute the following partial derivatives. The first k columns of C are given by $\delta_B \frac{\partial \lambda_B}{\partial \kappa^*} + \delta_L \frac{\partial \lambda_L}{\partial \kappa^*}$, where

$$\begin{aligned} \frac{\partial \lambda_B}{\partial \kappa^*} &= \frac{\kappa^* r_b \phi(\cdot) \Phi(\cdot) - r_b \rho_{v_B v_L} / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \phi(\cdot) \phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} &- \frac{\Phi_{ns}(\omega^* r_l, \rho_{v_B v_L} \kappa^* r_b, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ &= \lambda_B(\kappa^* r_b - r_b \rho_{v_B v_L} / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \frac{\phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} - \frac{\Phi_{ns}(\omega^* r_l, \rho_{v_B v_L} \kappa^* r_b, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]} \end{aligned}$$

because from normal distribution properties we have

$$\frac{\partial \phi(\kappa^* r_b)}{\partial \kappa^*} = \kappa^* r_b \phi(\kappa^* r_b),$$

$$\frac{\partial \Phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b)/(1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\partial \kappa^*} = \frac{\partial \Phi(t)}{\partial t} \frac{\partial t}{\partial \kappa^*} = \phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b)/(1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})r_l/(1 - \rho_{v_B v_L}^2),$$

$$\frac{\partial \Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}{\partial \kappa^*} = \frac{\partial}{\partial \kappa^*} \int_{-\infty}^{\omega^* r_l} [\int_{-\infty}^{\kappa^* r_b} \phi(a, b, \rho_{v_B v_L}) db] da = \int_{-\infty}^{\kappa^* r_b} \phi(\omega^* r_l, b, \rho_{v_B v_L}) db = \Phi_{ns}(\kappa^* r_b, \rho_{v_B v_L} \omega^* r_l, (1 - \rho_{v_B v_L}^2)),$$

where $\Phi_{ns}(\cdot)$ is a non-standardized normal cdf. Following the same rules we have

$$\begin{split} \frac{\partial \lambda_L}{\partial \kappa^*} &= \frac{r_b / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \phi(\cdot) \phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} - \frac{\Phi_{ns}(\omega^* r_l, \rho_{v_B v_L} \kappa^* r_b, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ &= \lambda_L (r_b / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \frac{\phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} - \frac{\Phi_{ns}(\omega^* r_l, \rho_{v_B v_L} \kappa^* r_b, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]}). \end{split}$$

The second k columns of C are given by $\delta_B \frac{\partial \lambda_B}{\partial \omega^*} + \delta_L \frac{\partial \lambda_L}{\partial \omega^*}$, where

$$\begin{split} \frac{\partial \lambda_B}{\partial \omega^*} &= \frac{r_l / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \phi(\cdot) \phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} - \frac{\Phi_{ns}(\kappa^* r_b, \rho_{v_B v_L} \omega^* r_l, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ &= \lambda_B (r_l / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \frac{\phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} - \frac{\Phi_{ns}(\kappa^* r_b, \rho_{v_B v_L} \omega^* r_l, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]}), \end{split}$$

and

$$\frac{\partial \lambda_L}{\partial \omega^*} = \frac{\omega^* r_l \phi(\cdot) \Phi(\cdot) - r_l \rho_{v_B v_L} / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \phi(\cdot) \phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} - \frac{\Phi_{ns}(\kappa^* r_b, \rho_{v_B v_L} \omega^* r_l, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ = \lambda_L (\omega^* r_l - r_l \rho_{v_B v_L} / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} \frac{\phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} - \frac{\Phi_{ns}(\kappa^* r_b, \rho_{v_B v_L} \omega^* r_l, (1 - \rho_{v_B v_L}^2)) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2}$$

The last column of C are given by $\delta_B \frac{\partial \lambda_B}{\partial \rho_{v_B v_L}} + \delta_L \frac{\partial \lambda_L}{\partial \rho_{v_B v_L}}$ where

$$\begin{aligned} \frac{\partial \lambda_L}{\partial \rho_{v_B v_L}} &= \frac{\rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{-\frac{1}{2}} [(1 - \rho_{v_B v_L}^2)^{-1} (\omega^* r_l - \kappa^* r_b \rho_{v_B v_L}) - \kappa^* r_b 1 / \rho_{v_B v_L}] \phi(\cdot) \phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} \\ &- \frac{\phi^2 (\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L}) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ &= \lambda_L (\rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{-\frac{1}{2}} [(1 - \rho_{v_B v_L}^2)^{-1} (\omega^* r_l - \kappa^* r_b \rho_{v_B v_L}) - \kappa^* r_b 1 / \rho_{v_B v_L}] \frac{\phi((\kappa^* r_b - \rho_{v_B v_L} \omega^* r_l) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} \\ &- \frac{\phi^2 (\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}{[\Phi^2(\cdot)]}), \end{aligned}$$

given that

$$\begin{aligned} \frac{\partial \Phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\partial \rho_{v_B v_L}} &= \frac{\partial \Phi(t)}{\partial t} \frac{\partial t}{\partial \rho_{v_B v_L}} \\ = & \phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}}) \rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}} [\omega^* r_l (1 - \rho_{v_B v_L}^2)^{-1} - \kappa^* r_b \frac{1}{\rho_{v_B v_L}} - \kappa^* r_b \rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{-1}], \\ & \frac{\partial \Phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}{\partial \rho_{v_B v_L}^2} = \phi^2(\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L}), \end{aligned}$$

from Plackett (1945), and

$$\begin{aligned} \frac{\partial \lambda_L}{\partial \rho_{v_B v_L}} &= \frac{\rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{-\frac{1}{2}} [(1 - \rho_{v_B v_L}^2)^{-1} (\kappa^* r_b - \omega^* r_l \rho_{v_B v_L}) - \omega^* r_l 1 / \rho_{v_B v_L}] \phi(\cdot) \phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi^2(\cdot)} \\ &- \frac{\phi^2 (\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L}) \phi(\cdot) \Phi(\cdot)}{[\Phi^2(\cdot)]^2} \\ &= \lambda_L (\rho_{v_B v_L} (1 - \rho_{v_B v_L}^2)^{-\frac{1}{2}} [(1 - \rho_{v_B v_L}^2)^{-1} (\kappa^* r_b - \omega^* r_l \rho_{v_B v_L}) - \omega^* r_l 1 / \rho_{v_B v_L}]}{\Phi(\cdot)} \frac{\phi((\omega^* r_l - \rho_{v_B v_L} \kappa^* r_b) / (1 - \rho_{v_B v_L}^2)^{\frac{1}{2}})}{\Phi(\cdot)} \\ &- \frac{\phi^2 (\kappa^* r_b, \omega^* r_l, \rho_{v_B v_L})}{[\Phi^2(\cdot)]}). \end{aligned}$$

Plugging in the consistently estimated parameters allow us to have consistent standard errors as well.

Semiparametric Estimation

So far we assumed that the joint distribution of unobservables is normal. Nevertheless, in financial phenomena it is sometimes hard to assume shocks' normality. In this section we outline a procedure to control for selectivity without imposing any distributional assumption. The asymptotic properties of the two-step estimator for semiparametric sample selection models have been derived by Newey (2009). His estimator works as the theoretical basis for ours. A notable application of semiparametric methods with multi-choice selection is Dahl (2002).

From system (15) without the normality assumption we have that

$$E[p_{lb}|s_b = 1, s_l = 1] = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + \psi(m_{lb}),$$
(26)

where

$$\psi(m_{lb}) = E(\epsilon_{lb}|s_b = 1, s_l = 1) = E(\epsilon_{lb}|\omega, \kappa, r_l, r_b)$$

is an unknown function. Thus, equation (26) implies that the mean of the outcome disturbances depends only on $m_{lb} = m(\omega, \kappa, r_l, r_b)$ conditional on the selection process, where m(.) is an unknown function.³² If we assume normality, the function $\psi(m_{lb})$ becomes the multivariate inverse Mills ratio. The term is a generalization of the correction term considered by Heckman and Robb (1985). Let us define $\tau(m, \eta)$ as a strictly monotonic transformation of each entry of the index m, depending on the parameter η . Let $P^K(\tau) = (P_{1K}(\tau), \ldots, P_{KK}(\tau))'$ be a vector of functions such that for large values of K a linear combination of $P^K(\tau)$ can approximate an unknown function of $\tau(\cdot)$. Let $\hat{\tau}_{lb} = \tau(\hat{m}_{lb}, \hat{\eta})$ and $\hat{p}_i = P^K(\hat{\tau}_{lb})$. Let us assume also that the approximating functions are power series given by $P_{kK}(\tau) = \tau^{k-1}$.³³ Thus, we can write model (26) as

$$E[p_{lb}|s_b = 1, s_l = 1] = \beta_0 + \beta_1 x_{lb} + \alpha q_{lb} + \sum_{k=1}^q \gamma_k \tau_{lb}^{k-1}.$$
(27)

In practice, to consistently estimate the parameters of this equation, we obtain $\hat{\tau}_{lb}$ from the first step using a fully parametric specification or distribution-free estimators that are available in the literature, including those of Manski (1975), Cosslett (1983), Powell et al. (1989), Ichimura (1993), Klein and Spady (1993) and Khan (2013). We then plug in those estimates in the second step, approximating the unknown conditional expected value of disturbances. See Newey (2009) for the asymptotics and the standard errors computation of such estimators.

³²This restriction is implied by the assumption of independence between disturbances and regressors.

³³Other approximating functions can be used. Spline approximation can be used as approximating functions. See, e.g. Newey (2009).

Appendix C: Estimation with False Positive

The data used in this paper are actually the output from an algorithm, as detailed in Arciero et al. (2016). To test the accuracy of their algorithm, they compare the output of this algorithm to a variety of statistics on unsecured interbank lending. They estimate that the type I error of the algorithm is quite low, meaning that the algorithm captures almost all of the interbank loans settled over TARGET2. The type II error is estimated to be larger, meaning that the algorithm is finding more interbank loans. Armantier and Copeland (2012) find similar results when testing a similar algorithm using U.S. payments data. In what follows we outline the assumptions needed for consistency using data from this algorithm. To simplify the notation, let us suppress the bilateral subscript lb and consider only the covariates. The rate equation in vector terms then becomes

$$P = \beta_0 + \beta_1 X + \epsilon, \tag{28}$$

Suppose loans can be split in true and false, then $P = [P'_T; P'_F]'$ and $X = [X'_T; X'_F]'$ and

$$P_T = \beta_{0,T} + \beta_{1,T} X_T + \epsilon_T, \tag{29}$$

$$P_F = \beta_{0,F} + \beta_{1,F} X_F + \epsilon_F, \tag{30}$$

For $\hat{\beta}_{OLS}$ to be a consistent estimator of $\beta_T = [\beta'_{0,T}; \beta'_{0,F}]'$ we need the following assumptions.

A1
$$\beta_F = \beta_T$$

A2 $E(X_T\epsilon_T) = E(X_F\epsilon_F) = 0$

If the algorithm is randomly picking false loans across pairs of banks, it is plausible to think that the relationship between X, Y and ϵ is not structurally different between the true loans subpopulation and the whole sample. Nevertheless, A1 and A2 are less strong assumptions. Under these assumptions, picking systematically a pair of banks in the sample, when they do not exchange money, does not imply a bias per se. A1 and A2 allow for systematic inclusion of pairs of banks in the sample as long as they are associated with random rates. Bias emerges when the pair is systematically wrongly included and associated with non random rates. For example if bank A and bank B are systematically included with high rates. If A and B are wrongly included but rates are drawn randomly, $\hat{\beta}_{OLS}$ is not biased.³⁴ From our viewpoint, it is difficult to find a priori an argument for false loans systematically associating rates and banks characteristics in a biased way. We thus believe that A1 and A2 are plausible in our context. Furthermore, we aggregate loans across the maintenance period, thus our variables are less prone to measurement error. In the empirical analysis we also show that selection bias is not a big issue for our sample, an evidence that supports randomness in measurement error. In addition, the estimated coefficients are in line with the findings in other studies, see Section 6.

³⁴In such a case only standard errors are upward biased, but this eventually works against finding significant marginal effects, thus leaning on the safe side.

Appendix D: MP1 and MP2 Selection Equations

Tables D.1 - D.4 describe the first steps, showing the results for the likelihood to trade as a borrower or lender -i.e. the selection of the counterparty-. This is the information that we use to control for the selectivity bias in rates and quantities. For the sake of brevity we comment only on Tables D.1 and D.2, which report respectively on the borrower and lender selection equations in MP1. The probability of engaging the market is modeled with a probit link as described in Appendix B.³⁵

From Table D.1 we can see that having borrowed from a higher number of counterparts increases the probability of being a borrower significantly. Big and well-capitalized banks are more likely to borrow money. Having more fixed (and thus less liquid) assets makes more likely a bank borrow in the money market. On average, it is more likely to observe Italian borrowers, while it is less likely that they are French, Spanish, Dutch, Irish, English, Belgian or from outside the EU.³⁶

On the supply side, Table D.2 reports that banks are more likely to lend if they have lent to an higher number of counterparts and less likely if they lent more in the past. Banks with more deposits and short-term funding or other interest bearing liabilities are less likely to operate as lenders. Nationality seems to matter less, only Italian lenders are more frequent while French ones are less.

 $^{^{35}\}mathrm{Results}$ obtained with a bivariate probit are almost identical to the ones from two independent probit estimates.

³⁶The reference country is Germany.

Dependent Variable: borrower bilatera	il trade						
Borrowing rate at $t-1$		-0.2623					
		(0.5147)					
Borrowed value at $t-1$		0.000					
Borrower's counterparts at $t = 1$		(0.0000)					
Borrower's counterparts at $t = 1$	(0.0021)						
	Own	Counterpart					
A loan	-0.0042**	-0.0035***					
ii iouii	(0.0012)	(0.0013)					
A fix as	0.1677***	0.0664**					
	(0.0278)	(0.0279)					
A non ern	0.0001	-0.0064*					
	(0.0039)	(0.0039)					
L dep sh fun	0.0023	-0.0051					
	(0.0045)	(0.0045)					
L oth int bea	-0.0020	-0.0054					
	(0.0046)	(0.0047)					
L oth res	0.0087	-0.0123					
T	(0.0147)	(0.0147)					
L equ	(0.0249^{++++})	-0.0024					
A tot accet	(0.0002)	0.0002)					
A tot asset	(0.0037)	(0.0001)					
IT	0.0194^{***}	0.006027					
11	(0.0009)	(0.0010)					
\mathbf{FR}	-0.0100***	0.0019					
	(0.0015)	(0.0015)					
ES	-0.0047***	-0.0021*					
	(0.0012)	(0.0012)					
NL	-0.0065***	-0.0010					
	(0.0013)	(0.0013)					
GR	-0.0016	0.0015					
	(0.0016)	(0.0016)					
IE	-0.0080***	-0.0060***					
	(0.0023)	(0.0023)					
UK	-0.0097	-0.0027 (0.0021)					
US/IAP/EX	-0.0071***	-0.0010					
05/341/114	(0.0016)	(0.0016)					
АТ	0.0009	0.0018					
	(0.0011)	(0.0012)					
PT	0.0005	-0.0014					
	(0.0014)	(0.0014)					
LU	-0.0030	-0.0015					
	(0.0022)	(0.0022)					
CY	0.0046^{**}	0.0009					
	(0.0022)	(0.0022)					
СН	-0.0148***	0.0012					
DI	(0.0029)	(0.0029)					
F1	-0.0024	0.0008					
FUEY	(0.0023)	0.0023)					
EUEA	-0.0033	-0.0003					
BE	-0.0093***	-0.0001					
	(0.0022)	(0.0023)					
Constant	(0.0022)	-0.0328***					
		(0.0071)					
		· /					
Time interval		2010-01-20 - 2010-02-09					
Maturity		1 to 3 days					
Observations		124962					

Table D.1: Borrower selection equation MP1

Notes: *: p < 0.10; **: p < 0.05; ***: p < 0.01. Only country fixed effects with more than 1% of observations are included in the model. The time interval is a maintenance period, t - 1 refers to the previous time interval. A stands for assets, L for liabilities. Country fixed effects are reported using the usual labels, EX means other foreign countries w.r.t. the eurozone, EUEX means other countries in the eurozone that are not included with individual fixed effects.

Lending rate at $t-1$	0.1335		
T . 1	(0.4743)		
Lent value at $t - 1$		-0.0001***	
Lender's counterparts at $t = 1$		(0.0000) 0.0020***	
Equation 5 counterparts at $i = 1$		(0.0004)	
	Own	Counterpart	
A loan	0.0012	-0.0035***	
	(0.0013)	(0.0013)	
A fix as	-0.0453	0.1184 ***	
	(0.0291)	(0.0277)	
A non ern	-0.0047	-0.0048	
	(0.0039)	(0.0039)	
L dep sh fun	-0.0111**	-0.0029	
	(0.0045)	(0.0045)	
L otn int bea	-0.0131^{+++}	-U.UU8U	
oth ros	0.0040)	(0.0040)	
L oth res	-0.0228	0.0000	
L equ A tot asset	-0.0077	0.0140)	
	(0.0062)	(0.0091)	
	-0.0002)	0.0002)	
1 000 00000	(0.0002)	(0.0020)	
T	0.0075 ***	0.0209 ***	
-	(0.0010)	(0.0009)	
FR	-0.0042***	-0.0003	
	(0.0015)	(0.0015)	
ES NL GR	-0.0000	-0.0009	
	(0.0012)	(0.0012)	
	-0.0009	-0.0019	
	(0.0013)	(0.0013)	
	0.0021	0.0031 *	
	(0.0016)	(0.0016)	
ΙE	-0.0015	-0.0039*	
	(0.0023)	(0.0023)	
UK	0.0018	-0.0060***	
	(0.0021)	(0.0021)	
US/JAP/EX	-0.0001	-0.0053***	
	(0.0016)	(0.0016)	
AT	0.0009	0.0054^{***}	
PT	(0.0012)	(0.0011)	
	(0.0005)	0.0030^{mm}	
LU	0.0014)	(0.0014) _0.0024	
	(0.0032)	-0.0024 (0.0029)	
T Y	0.0022)	-0.0022)	
	(0.0022)	(0.0022)	
CH FI	0.0009	-0.0005	
	(0.0029)	(0.0029)	
	0.0007	-0.0021	
	(0.0023)	(0.0022)	
EUEX BE	0.0001	-0.0004	
	(0.0009)	(0.0009)	
	0.0007	-0.0071***	
	(0.0022)	(0.0022)	
Constant	× /	-0.0075	
	(0.0071)		
Γime interval	2010-01-20 - 2010-02-09		
Maturity		1 to 3 days	
	124962		

Table D.2: Lender selection equation MP1

Notes: See Table D.1.

Dependent Variable: borrower bilateral trade at time t				
		1 6405 ***		
Borrowing rate at $t-1$		-1.6495 *** (0.2030)		
Borrowed value at $t-1$		0.0000		
		((0.0000))		
Borrower's counterparts at $t-1$		0.0088 ***		
		(0.0007)		
A loop	Own	Counterpart		
A IOan	(0.0016)	-0.0003		
A fix as	0.0301	-0.0009		
	(0.0256)	(0.0244)		
A non ern	0.0042	-0.0022		
	(0.0040)	(0.0040)		
L dep sh fun	0.0032	0.0025		
L oth int has	(0.0045) 0.0021	(0.0045)		
L oth mt bea	(0.0021)	(0.0001)		
L oth res	0.0144	-0.0005		
	(0.0167)	(0.0166)		
L equ	0.0079	0.0030		
	(0.0063)	(0.0062)		
A tot asset	0.0010 ***	0.0023 ***		
IT.	(0.0002)	(0.0002)		
11	(0.0014)	(0.0191)		
FB	-0.0039 **	0.0019		
110	(0.0016)	(0.0016)		
ES	0.0002	0.0028 **		
	(0.0013)	(0.0013)		
NL	0.0009	-0.0028 *		
CP	(0.0015)	(0.0015)		
GR	(0.0000)	(0.0003)		
ΙE	0.0039	0.0065 ***		
	(0.0025)	(0.0024)		
UK	-0.0090 ***	-0.0073 ***		
/ /	(0.0023)	(0.0023)		
US/JAP/EX	-0.0012	-0.0038 **		
۸T	(0.0017)	(0.0017)		
AI	(0.0028)	(0.0012)		
РТ	0.0024	0.0004		
	(0.0016)	(0.0016)		
LU	0.0002	-0.0007		
	(0.0024)	(0.0024)		
CY	0.0010	-0.0016		
FUEX	(0.0024)	(0.0024)		
LUEA	(0.0031)	(0.0031)		
FI	0.0029	-0.0038		
	(0.0024)	(0.0024)		
EUEX	0.0007	0.0018 *		
	(0.0010)	(0.0010)		
BE	-0.0007	-0.0095 *** (0.0024)		
Constant	(0.0024)	(0.0024) -0.0377 ***		
Silvan		(0.0072)		
Time interval	2	2009-02-11 - 2009-03-10		
Maturity		1 to 3 days		
Observations		123552		

Table D.3: Borrower selection equation MP2

Notes: See Table D.1.

Lending rate at $t-1$	0.4553		
T , 1 , 1 , 1	(0.2817)		
Lent value at $t - 1$		(0.0000)	
Lender's counterparts at $t-1$		0.0011	
r in the second s		(0.0009)	
	Own	Counterpart	
A loan	-0.0016	-0.0041 ***	
A for ac	(0.0015)	(0.0014)	
A IIX as	(0.0024)	$(0.2570^{-0.12})$	
A non ern L dep sh fun	-0.0015	0.0007	
	(0.0040)	(0.0040)	
	0.0071	0.0090 **	
	(0.0046)	(0.0046)	
L oth int bea	0.0064	0.0114 **	
1	(0.0047)	(0.0047)	
L oth res	(0.0169)	(0.0167)	
L equ	0.0168)	(0.0167) 0.0357 ***	
	(0.0052)	(0.0063)	
A tot asset IT	0.0006 ***	0.0047 ***	
	(0.0002)	(0.0002)	
	0.0041 ***	0.0129 ***	
	(0.0011)	(0.0010)	
FR	0.0007	-0.0107 ***	
ES	(0.0017)	(0.0016)	
	-0.0002	$-0.0066^{+0.0}$	
NT	(0.0014)	-0.00013)	
	(0.0015)	(0.0015)	
GR	-0.0001	-0.0056 ***	
	(0.0018)	(0.0018)	
E	0.0046 *	-0.0075 ***	
	(0.0025)	(0.0025)	
JK	-0.0026	-0.0156 ***	
IS/IAD/FY	(0.0023)	(0.0023)	
JS/JAF/EA	(0.0017)	(0.0017)	
ΥТ	0.0009	-0.0019	
	(0.0012)	(0.0012)	
PT	0.0028 *	-0.0039 **	
	(0.0016)	(0.0016)	
LU	-0.0019	-0.0038	
	(0.0024)	(0.0024)	
JI	-0.0026	0.0036	
EUEX	-0.0023	-0.0199 ***	
	(0.0031)	(0.0031)	
FI	-0.0005	-0.0014	
	(0.0025)	(0.0024)	
EUEX	0.0010	-0.0049 ***	
	(0.0010)	(0.0010)	
3E	-0.0025	-0.0001	
Q	(0.0024)	(0.0023)	
onstant	-0.0629 ***		
		(0.0072)	
lime interval	2009-02-11 - 2009-03-10		
Maturity	2000	1 to 3 davs	
Observations	123552		
	120002		

Table D.4: Lender selection equation MP2

Notes: See Table D.1.

Appendix E: Additional Results on Trading Patterns during the Sovereign Crisis (Section 6.3.2) Probability to Trade



Figure E.1: Probability to trade - Lender's country dummies.

x 10³ Lender BE

0

-5

Notes: See Figure 1, Figure 4, Figure 9 and 10.



Figure E.2: Probability to trade - Borrower's country dummies.

Notes: See Figure 1, Figure 4, Figure 9 and 10.

Quantity Exchanged

Figure E.3: Quantity equation - Lender and borrower balance sheet covariates.

Notes: See Figure 4 and 15.

Figure E.4: Quantity equation - Borrower country dummies.

Notes: See Figure 1, Figure 4, Figure 9 and 10.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1129 *The effects of central bank's verbal guidance: evidence from the ECB*, by Maddalena Galardo and Cinzia Guerrieri (July 2017).
- N. 1130 The Bank of Italy econometric model: an update of the main equations and model elasticities, by Guido Bulligan, Fabio Busetti, Michele Caivano, Pietro Cova, Davide Fantino, Alberto Locarno, Lisa Rodano (July 2017).
- N.1131 Venture capitalists at work: what are the effects on the firms they finance?, by Raffaello Bronzini, Giampaolo Caramellino and Silvia Magri (September 2017).
- N.1132 Low frequency drivers of the real interest rate: a band spectrum regression approach, by Fabio Busetti and Michele Caivano (September 2017).
- N. 1133 *The real effects of relationship lending*, by Ryan Banerjee, Leonardo Gambacorta and Enrico Sette (September 2017).
- N.1134 *Credit demand and supply: a two-way feedback relation*, by Ugo Albertazzi and Lucia Esposito (September 2017).
- N. 1135 Legislators' behaviour and electoral rules: evidence from an Italian reform, by Giuseppe Albanese, Marika Cioffi and Pietro Tommasino (September 2017).
- N.1136 Macroeconomic effects of non-standard monetary policy measures in the euro area: the role of corporate bond purchases, by Anna Bartocci, Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (September 2017).
- N. 1137 On secular stagnation and low interest rates: demography matters, by Giuseppe Ferrero, Marco Gross and Stefano Neri (September 2017).
- N. 1138 Tony Atkinson and his legacy, by Rolf Aaberge, François Bourguignon, Andrea Brandolini, Francisco H. G. Ferreira, Janet C. Gornick, John Hills, Markus Jäntti, Stephen P. Jenkins, Eric Marlier, John Micklewright, Brian Nolan, Thomas Piketty, Walter J. Radermacher, Timothy M. Smeeding, Nicholas H. Stern, Joseph Stiglitz and Holly Sutherland (September 2017).
- N. 1139 *Credit misallocation during the European financial crisis*, by Fabiano Schivardi, Enrico Sette and Guido Tabellini (September 2017).
- N. 1140 Natural rates across the Atlantic, by Andrea Gerali and Stefano Neri (September 2017).
- N.1141 A quantitative analysis of risk premia in the corporate bond market, by Sara Cecchetti (October 2017).
- N. 1142 *Monetary policy in times of debt*, by Mario Pietrunti and Federico M. Signoretti (October 2017).
- N.1143 *Capital misallocation and financial development: a sector-level analysis*, by Daniela Marconi and Christian Upper (October 2017).
- N. 1144 Leaving your mamma: why so late in Italy?, by Enrica Di Stefano (October 2017).
- N. 1145 A Financial Conditions Index for the CEE economies, by Simone Auer (October 2017).
- N. 1146 Internal capital markets in times of crisis: the benefit of group affiliation in Italy, by Raffaele Santioni, Fabio Schiantarelli and Philip E. Strahan (October 2017).
- N. 1147 *Consistent inference in fixed-effects stochastic frontier models*, by Federico Belotti and Giuseppe Ilardi (October 2017).
- N. 1148 Investment decisions by European firms and financing constraints, by Andrea Mercatanti, Taneli Mäkinen and Andrea Silvestrini (October 2017).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBERTAZZI U., G. ERAMO, L. GAMBACORTA and C. SALLEO, Asymmetric information in securitization: an empirical assessment, Journal of Monetary Economics, v. 71, pp. 33-49, TD No. 796 (February 2011).
- ALESSANDRI P. and B. NELSON, *Simple banking: profitability and the yield curve,* Journal of Money, Credit and Banking, v. 47, 1, pp. 143-175, **TD No. 945 (January 2014).**
- ANTONIETTI R., R. BRONZINI and G. CAINELLI, *Inward greenfield FDI and innovation*, Economia e Politica Industriale, v. 42, 1, pp. 93-116, **TD No. 1006 (March 2015).**
- BARONE G. and G. NARCISO, Organized crime and business subsidies: Where does the money go?, Journal of Urban Economics, v. 86, pp. 98-110, **TD No. 916 (June 2013).**
- BRONZINI R., The effects of extensive and intensive margins of FDI on domestic employment: microeconomic evidence from Italy, B.E. Journal of Economic Analysis & Policy, v. 15, 4, pp. 2079-2109, TD No. 769 (July 2010).
- BUGAMELLI M., S. FABIANI and E. SETTE, The age of the dragon: the effect of imports from China on firmlevel prices, Journal of Money, Credit and Banking, v. 47, 6, pp. 1091-1118, TD No. 737 (January 2010).
- BULLIGAN G., M. MARCELLINO and F. VENDITTI, *Forecasting economic activity with targeted predictors,* International Journal of Forecasting, v. 31, 1, pp. 188-206, **TD No. 847 (February 2012).**
- BUSETTI F., On detecting end-of-sample instabilities, in S.J. Koopman, N. Shepard (eds.), Unobserved Components and Time Series Econometrics, Oxford, Oxford University Press, **TD No. 881 (September 2012).**
- CESARONI T., *Procyclicality of credit rating systems: how to manage it,* Journal of Economics and Business, v. 82. pp. 62-83, **TD No. 1034 (October 2015).**
- CIARLONE A., *House price cycles in emerging economies*, Studies in Economics and Finance, v. 32, 1, **TD No. 863 (May 2012).**
- CUCINIELLO V. and F. M. SIGNORETTI, *Large bank, loan rate markup and monetary policy,* International Journal of Central Banking, v. 11, 3, pp. 141-177, **TD No. 987 (November 2014).**
- DE BLASIO G., D. FANTINO and G. PELLEGRINI, *Evaluating the impact of innovation incentives: evidence from an unexpected shortage of funds*, Industrial and Corporate Change, v. 24, 6, pp. 1285-1314, **TD No. 792 (February 2011).**
- DEPALO D., R. GIORDANO and E. PAPAPETROU, *Public-private wage differentials in euro area countries:* evidence from quantile decomposition analysis, Empirical Economics, v. 49, 3, pp. 985-1115, **TD No. 907 (April 2013).**
- DI CESARE A., A. P. STORK and C. DE VRIES, *Risk measures for autocorrelated hedge fund returns*, Journal of Financial Econometrics, v. 13, 4, pp. 868-895, **TD No. 831 (October 2011).**
- FANTINO D., A. MORI and D. SCALISE, Collaboration between firms and universities in Italy: the role of a firm's proximity to top-rated departments, Rivista Italiana degli economisti, v. 1, 2, pp. 219-251, TD No. 884 (October 2012).
- FRATZSCHER M., D. RIMEC, L. SARNOB and G. ZINNA, *The scapegoat theory of exchange rates: the first tests*, Journal of Monetary Economics, v. 70, 1, pp. 1-21, **TD No. 991 (November 2014).**
- NOTARPIETRO A. and S. SIVIERO, *Optimal monetary policy rules and house prices: the role of financial frictions,* Journal of Money, Credit and Banking, v. 47, S1, pp. 383-410, **TD No. 993 (November 2014).**
- RIGGI M. and F. VENDITTI, *The time varying effect of oil price shocks on euro-area exports,* Journal of Economic Dynamics and Control, v. 59, pp. 75-94, **TD No. 1035 (October 2015).**
- TANELI M. and B. OHL, *Information acquisition and learning from prices over the business cycle,* Journal of Economic Theory, 158 B, pp. 585–633, **TD No. 946 (January 2014).**

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, *Local development that money cannot buy: Italy's Contratti di Programma,* Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, *Stock market efficiency in China: evidence from the split-share reform*, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, **TD No. 969** (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**
- ESPOSITO L., A. NOBILI and T. ROPELE, *The management of interest rate risk during the crisis: evidence from Italian banks*, Journal of Banking & Finance, v. 59, pp. 486-504, **TD No. 933 (September 2013).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, *Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility,* Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, **TD No. 896 (January 2013).**
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, TD No. 968 (July 2014).

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**

- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010)**.
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

FORTHCOMING

- ADAMOPOULOU A. and E. KAYA, Young Adults living with their parents and the influence of peers, Oxford Bulletin of Economics and Statistics, **TD No. 1038 (November 2015).**
- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis,* Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, **TD No. 1077 (July 2016).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).